THE ROLE OF HEALTH PROMOTION WITHIN INTEGRATED HEALTH SYSTEMS

A POSITION PAPER BY THE CENTRE FOR HEALTH PROMOTION

JANUARY 1998

Produced by the Working Group on Integrated Health
Systems
Lead Author: Elizabeth Birse

ACKNOWLEDGEMENTS

This paper is the most substantial of many tasks the Centre's Working Group on Integrated Health Systems has pursued. The Working Group's first task was sponsoring a well attended and very successful forum on the topic. At that time, a number of activities were undertaken, with major emphasis on the following paper.

Great thanks are due to Elizabeth Birse for her lead authorship of this paper. We feel the paper is a significant piece of scholarship which adds significantly to the field of Health Promotion. In addition to the research and writing aspects, Elizabeth was very effective is seeking and incorporating specific and substantial feedback from an advisory committee.

Great thanks are also due to the former East York Board of Health who provided funds so that this paper could be written.

Finally, great thanks are due to Committee members who contributed to the overall design of the paper, as well as providing feedback to earlier drafts, both at meetings and in individual dialogue and correspondence with the lead author, Elizabeth Birse.

Irving Rootman, Director, Centre for Health Promotion

Larry Hershfield, Manager, The Health Communication Unit

WORKING GROUP ON INTEGRATED HEALTH SYSTEMS—MEMBERS

Shelley Adams, Waterloo Region DHC Kimberley Badovinac, Ontario Public Health Association Sheela Basrur, East York Health Unit Elizabeth Birse, Quantum Solutions Sandy Bollenbach, City of York Health Unit Jenny Carryer Neville Chenoy Michael Finkelstein

Michael Goodstadt, Centre for Health Promotion

Heather Graham, The Doctors Hospital

Maria Herrera, City of Toronto Public Health Department, Health Promotion & Advocacy

Larry Hershfield, Centre for Health Promotion, Project Office

Marla Jackson, Four Villages Community Health Centre

Marg Muir, Sunnybrook Health Science Centre

Mary Frances O'Hagan, Institute of Human Potential

Irving Rootman, Centre for Health Promotion

Colleen Stanton

David Vickers, Health Economic Development, Ontario Ministry of Health

Dianna Vidovic, Centre for Health Promotion, Project Office

Lori Wilson, Metro Toronto District Health Council

TABLE OF CONTENTS

Executive Summary	Page 1
Background Statement	Page 4
Introduction	Page 6
I: The Primary Goal of IHS: The Promotion of Health	
The Meaning of Health	Page 7
What is Health Promotion?	Page 7
Core Values	Page 8
Promoting Health: Three Complementary Approaches	Page 8
The Medical Approach	Page 9
The Lifestyle/Behavioural Approach	Page 10
The Socioenvironmental Approach	Page 11
IHS Accountability	Page 13
II. A I :fortale/Debosiessel Assume als to Health Duras at	
II: A Lifestyle/Behavioural Approach to Health Promotion	
Key IHS Strategies	
Health Education A Vision for IUS Implementation	•
Health Education: A Vision for IHS Implementation Health Communication	
Health Communication: A Vision for IHS	Page 10
	Dogo 21
ImplementationBrief Interventions	_
Brief Interventions: A Vision for IHS Implementation	•
Self-Help and Mutual Aid	_
Self-Help/Mutual Aid: A Vision for IHS	1 age 23
Implementation	Paga 26
Self-Care (Self-Guided Health Education)	_
Self-Care: A Vision for IHS Implementation	_
Healthy Public Policy	_
Healthy Public Policy: A Vision for IHS	1 age 2)
Implementation	Расе 31
1111p1v111v11tmtiv11	I ugo JI

III: A Socioenvironmental Approach to Health Promotion:

Key IHS Strategies	Page 32
Community Development	
Community Development: A Vision for IHS	-
Implementation	Page 35
Community Economic Development	Page 35
Community Economic Development: A Vision for	
IHS Implementation	Page 37
Healthy Public Policy and Health Advocacy	
Healthy Public Policy: A Vision for IHS	_
Implementation	Page 39
Health Advocacy: A Vision for IHS Implementation	_
Create Supportive Environments	_
Create Supportive Environments: A Vision for	C
IHS Implementation	Page 43
: Core Values and their Implications for IHS	Page 45
Empowerment	•
Public Participation	Page 50
Addressing the Impact of Broader Determinants	
of Health	Page 52
Reducing Social Inequities and Injustice	Page 53
Facilitating Intersectoral Collaboration	Page 56
onclusion	Page 61
uncusion	1 age 01
ppendix A: Suggested IHS Evaluation Indicators	Page 62
eferences	Page 65
31 CT CTTCCS	r age ().)

EXECUTIVE SUMMARY

The purpose of this paper is to inform the members of the Ministry of Health's Health Strategies Group about the critical role of health promotion within IHS reform in Ontario. It addresses, in turn, questions concerning the meaning of health and health promotion, what it means for IHSs to be held accountable for promoting the health of their members, evidence regarding the effectiveness of key health promotion strategies and their inclusion as core IHS' services or areas of activity, and the practical implications of embedding the values of health promotion into the design and operations of IHSs.

Health promotion theory and practice is grounded on the belief that health is far more than the absence of disease. Rather, it is a positive concept that emphasizes one's mental, social and spiritual well-being, and one's personal and social resources, in addition to one's physical well-being. In order to effectively enhance the health of Ontarians, our health system needs to be reoriented from its primary focus on treatment, rehabilitation, and the prevention of physiological risk factors, to a more balanced and comprehensive approach to promoting health. It is our position that this outcome can be successfully realized by guiding IHS reform according to the philosophy and values of health promotion, and by incorporating key health promotion strategies into the mandate of IHSs.

In order to most effectively influence the health outcomes of their clients and communities, IHSs must balance their attention and resources amongst a medical, lifestyle/behavioural, and a socioenvironmental approach to promoting health. These three approaches each target a specific set of health determinants. Therefore, IHSs will need to incorporate a variety of complementary health promotion strategies into their core basket of services and activities.

As the different approaches to health promotion address different health determinants, IHSs should be evaluated against a comprehensive set of indicators that measure the effectiveness of each approach, and the impact that each approach has on the health outcomes of their clients and communities. Among other factors, evaluation indicators should measure reductions in morbidity and mortality rates, physiological health risks, and high risk behaviours, as well as increases in quality of life and social well-being. Recognizing that these latter measures are beyond the sole control of both our health system and individual IHSs, IHSs should be held accountable for working collaboratively with others to address the broader determinants of health.

Of these three approaches to promoting health, this paper focuses on a lifestyle/behavioural approach and a socioenvironmental approach. A lifestyle/behavioural approach is concerned with the development of personal skills and public policies that support healthy choices and reduce individual risk factors. Strategies encompassed in this approach include health education, health

communication, brief interventions, self-help and mutual aid, self-care, and healthy public policy. Based on substantial evidence supporting the appropriateness and effectiveness of adopting each strategy to address all major risk factors across all levels of care, it is recommended that these strategies become core IHS services. Strategies should be supported by multidisciplinary teams of IHS' staff across all health delivery domains, particularly in the area of primary health. The feasibility of incorporating these health promotion strategies into the core business of IHSs is demonstrated through examples of possible IHS implementation within the community, diverse health settings, clients' homes, and other sites such as schools and workplaces. More comprehensive, multicomponent efforts are also recommended, and suggestions are made for potential collaboration between IHSs and a variety of community partners.

A socioenvironmental approach to health promotion addresses the health impacts of broader, socioenvironmental issues such as income, housing, social support etc.. Key strategies encompassed in this approach include community development, community economic development, healthy public policy, and creating environments that are supportive of health. Based on evidence regarding the contribution that each strategy can play in addressing the health impacts of these broader issues, it is recommended that they be incorporated as core areas of IHS activity. A portion of IHS' staff and other resources should be therefore be dedicated to this purpose. Examples are provided to illustrate how IHSs could foster and support collaborative initiatives with external health-related agencies and sectors at both the provincial and local levels. Of all potential partners, IHSs will need to work most closely with their communities, supporting members in identifying their own priority issues, and in working to positively influence those factors which affect their health and well-being.

In addition to incorporating key health promotion strategies into the mandate of IHSs, it is our position that the core values of health promotion are embedded in the design and operation of IHSs. Standards and guidelines that support the values of empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration are provided. Recommendations address:

- the empowerment of IHS' rostered communities, clients, agencies, organizations and employees;
- meaningful, representative public participation on IHS' governance bodies and committee structures, as well as in regard to specific programs and projects, including community health assessments;
- IHSs allocation of adequate resources to support efforts and collaborative partnerships that recognize and act upon the broader determinants of health;

- the reduction of social inequities and injustice through risk-adjusted funding
 formulas and special funding for programs that meet the needs of marginalized
 or high risk populations, and setting clear standards for IHS outreach efforts,
 committee structures, and efforts to remove systemic barriers to the
 determinants of health; and,
- collaboration amongst the component parts of the IHS (e.g. acute-, long-term-, and primary-care, health promotion, health-related social services), the adoption of a cooperative or corporate model of governance that represents the community served rather than the component provider parts of the system, core services and activities that IHSs should be accountable for providing, and issues related to intersectoral collaboration between IHSs and other agencies and sectors that have an impact on the broader determinants of health.

This paper suggests that, through its philosophy, strategies, and core values, the field of health promotion can and should play a critical role in IHS reform in Ontario. It is our hope that the Health Strategies Group at the Ministry of Health will understand and support our position, and will work in partnership with those who are and will be involved in IHS reform, to ensure that our health system is reoriented to promote and enhance the health of Ontarians. For our part, we are willing and eager to be involved in all IHS initiatives that are founded on the philosophy and values of health promotion, and that incorporate health promotion strategies into the core business of IHSs.

BACKGROUND STATEMENT

One of five key health promotion action areas identified in the Ottawa Charter is the reorientation of health services (WHO, 1986). A reoriented health care system has health, and not merely the absence of disease, as its primary outcome. To support prevention and promotion initiatives, increased attention and resources must be redirected from treatment and rehabilitation. Second, a reoriented health system emphasizes services located in communities, and relevant to their particular needs. This demands a shift from an "institutional" model of service delivery to strengthened "community-based" health services. It also demands a commitment to foster active public participation in the planning, development, operation and evaluation of health services. Lastly, the Charter describes a reoriented health system as operating according to a "health promotion" rather than a "service delivery" philosophy, such that the system's primary outcome is the enhancement of health at the individual and population levels.

Health reform presents a valuable opportunity to reorient Ontario's health care delivery sector according to the goals, objectives and values described in the Ottawa Charter. The field of health promotion supports the Ministry of Health's commitment to improve the health system's responsiveness, effectiveness, efficiency and accountability. We also support the Health Strategies Office in their efforts to develop a policy framework for Integrated Health Systems (IHSs). A draft of the framework describes an IHS as a nonprofit organization with a single point of accountability. The IHS is responsible for meeting a broad range of health needs for a defined group of Ontario residents (its roster) through the provision or purchase of a coordinated continuum of horizontally and vertically integrated services. Primary care is identified as the essential foundation of the system, which has an increased focus on health promotion and disease prevention. Needsbased capitated funding is reallocated by the IHS governance board in a way that best meets the needs of its members. Integrated information systems facilitate needs-based planning, evaluation, and coordination within each system and between IHSs and the Ministry of Health. The IHS operates under broad rules and standards set by the Ministry of Health in areas such as scope of and access to services, use of funds, and expectations for health outcomes/accountabilities (Health Strategies Group, 1997). Upon reviewing these key elements, the field of health promotion believes that Integrated Health Systems can be aligned with the goals, values, and practice of health promotion.

However, simply restructuring our traditional health care delivery organizations into an integrated system will not effectively the promote the health of Ontarians. In 1974, the Lalonde Report, entitled *A New Perspective on the Health of Canadians*, broke new ground by drawing attention to the fact that, on their own, traditional health care organizations are insufficient to achieve better health for Canadians (Lalonde, 1974). Up until this time, health status was largely considered to be the result of biology and, more importantly, access to a quality tertiary health care system (Hyndman, 1997). In addition to genetics and a formal health care system, the report identified lifestyles and environments as significant determinants of health (Lalonde, 1974). Despite these findings, the vast majority of health resources continue to be allocated to health care

organizations and providers with a medical focus on cure and treatment of disease. In Canada, approximately half of all health care expenditures are spent on institutional care (39% hospitals, 10% homes for special care), about one quarter on professional services (23%, of which physicians receive the bulk), and about 14% on drugs and appliances (Shah, 1994). The remaining 13% of our health budget covers the costs of research, health insurance administration, and public and voluntary health agencies, with less than 3.5% of all health care dollars spent on activities related to disease prevention and health promotion (Shah, 1994).

While restructuring may be an effective and efficient way to maintain the quality and reduce the overall costs of health care delivery, these efforts will not significantly improve our system's ability to enhance health outcomes. For this vision to be reached, Integrated Health Systems must be reoriented toward the promotion of health.

INTRODUCTION

Believing that IHSs have the potential to promote the health of Ontarians, the field of health promotion is prepared to support and contribute to their development. This position paper is intended to support IHS reform in two ways. First, it outlines our position on the direction of IHS reform and the role of health promotion within it. Secondly, it provides the Ministry of Health with our best thinking to date on practical ways in which to ensure that health promotion values and strategies are embedded in the planning, development and operations of IHSs.

Section I identifies the promotion of health as the primary goal or purpose of an IHS. It clarifies what is meant by *health*, defines the process of health promotion, and describes the core values and principles of health promotion, which we believe should guide IHS reform. Three distinct approaches to enhancing health are outlined: the medical approach to illness prevention; the individual/lifestyle approach to illness prevention and health promotion; and the socioenvironmental approach to health promotion. The integration of these three approaches within IHSs is identified as necessary to significantly improving the health outcomes of the populations served. Accountability for promotion of health is discussed, along with different types of indicators required to evaluate the effectiveness of IHSs in improving health outcomes. A list of suggested indicators can be found in Appendix A.

Section II identifies and describes the effectiveness of key health promotion strategies that address individual/lifestyle risk factors. The goal of these strategies is to assist individuals in making healthier choices through changing health related knowledge and attitudes, developing personal skills, and supporting healthy lifestyles through public policy. Inclusion of these strategies as core IHS services is supported through examples of how these strategies can be integrated throughout the IHS.

Section III describes and identifies the effectiveness of key health promotion strategies to address the impact of the socioenvironmental determinants of health. The goals of these strategies are to create environments that are supportive of health, to strengthen community action, and to develop healthy public policy that addresses the risk conditions in which people live and work. The inclusion of these strategies as core IHS services is supported with examples of how IHSs can work collaboratively with others to achieve these goals.

Section IV reviews the core values and principles that should guide IHS reform, and outlines how they should be reflected in the planning, development and operation of IHSs. Their incorporation into Ministry of Health standards and guidelines, is addressed, as are implications for recommended IHS strategies, structures, culture, and services.

The paper concludes with a summary of our position on the critical role that health promotion strategies and values can and should play within IHS reform.

I: THE PRIMARY GOAL OF IHSs: THE PROMOTION OF HEALTH

THE MEANING OF HEALTH

It is our position that the primary objective and outcome of Ontario's integrated health systems should be the promotion of health. This requires a common understanding of what is meant by health. At the root of health promotion theory and practice is the belief that health is "a positive concept emphasizing social and personal resources, as well as physical capacities" (WHO, 1986). Far more than the absence of disease or illness, health is the maximal attainment of physical, mental, social and spiritual well-being. This broad, holistic understanding of what it means to be healthy radically challenges our health sector's dominant, narrow definition of health. A health promotion perspective demands that we surface and test our assumptions of what it means to be healthy, how health is measured or assessed, the factors that determine health, and, most importantly, how we can most effectively employ our personal and public resources to enhance health at the individual, community and population level.

It is our position that these are the fundamental questions that need to be addressed in order to develop an effective policy framework for the development of IHSs that will in fact enhance health. This means that in addition to providing and coordinating health care services, integrated health systems must be concerned with, and actively promote, all aspects of health.

WHAT IS HEALTH PROMOTION?

Health promotion is "the process of enabling people to increase control over, and to improve, their health" and the issues that are identified as important health determinants (WHO, 1986). The overarching goal of health promotion is to achieve health for all, and to enhance individual and societal well-being. To move us toward this outcome, the field has developed an ethical and theoretical body of knowledge regarding the holism of health and its determinants (Rootman and Goodstadt, 1996). At a more practical level, this philosophical approach is supported by a wide variety of tools and techniques that may be categorized according to the five action areas identified in the Ottawa Charter: developing of personal skill; strengthening community action; creating and supporting healthy environments; building healthy public policy; and reorienting health services. Health promotion strategies are complementary and effective at the individual, organizational and community levels.

It is our position that founding IHS reform on the philosophy of health promotion, and incorporating health promotion strategies into the core business of IHSs, we will succeed in reorienting our health system and enhancing the health of Ontarians.

CORE VALUES

The process of health promotion is characterized by a set of core values or principles that guide all efforts to enhance health and support individual and societal well-being. Underlying all health promotion initiatives is the concept of *empowerment*. Empowerment refers to the capacity of individuals and communities to improve their health by increasing their control over the determinants that are important to their health.

Secondly, health promotion emphasizes active *public participation* in processes that encourage and enable individuals and communities to define, analyze and act upon events affecting their lives and living conditions (Labonte, 1993, 5).

Recognizing that health is influenced by more than genetics, individual lifestyles, and the provision of a health care delivery system, the field is committed to *addressing the impact of broader determinants of health*. This requires that health promotion interventions recognize and act upon the socioenvironmental conditions that shape the world in which we live.

Fourth, health promotion is committed to *reducing social inequities and injustice*, such that every individual, family and community may benefit from living, learning and working in a health-supporting environment.

Finally, health promotion seeks to *facilitate intersectoral collaboration* and initiate coordinated efforts to promote individual and community health.

It is our position that these same core values should guide the planning, development and operation of IHSs. This assertion is further developed in Section IV of this paper.

PROMOTING HEALTH: THREE COMPLEMENTARY APPROACHES

Our definition of health shapes our assumptions about its determinants, and the approaches we adopt in an effort to promote health. It also determines the indicators we use to measure how effective our efforts have been. Outlined below are three distinct approaches to enhance or promote health: medical; lifestyle or behavioural; and socioenvironmental (Labonte, 1992). Each approach targets a specific set of health determinants through a variety of strategies or interventions. As each approach is distinct, its effectiveness must be evaluated against a unique set of health indicators. In isolation, each approach is insufficient to significantly impact the health of Ontarians. When integrated, however, they provide a comprehensive approach to promoting health. IHSs must balance their attention and resources between our current focus on treatment, rehabilitation, and the prevention of physiological risk factors, with strategies targeting individual risk factors, and socioenvironmental risk conditions.

THE MEDICAL APPROACH

THE MEDICAL APPROACH TO PROMOTING HEALTH: AN OVERVIEW

Traditionally, our health services system has understood health to be the absence of disease or illness. Therefore, the primary focus has been on curing, treating, and, more recently, preventing disease and illness. Targeted risk factors tend to be medically or physiologically defined. Preventive medical efforts are directed toward individuals whose genetic, behavioural, personal or family history places them at greater risk of developing a serious disease or disability (Labonte, 1997). While the medical approach improves or promotes health through the prevention of disease, it is not a "health promotion" approach in that it is directed toward specific diseases rather than a positive concept of health. This distinction holds true for treatment and rehabilitation efforts which are directed toward the care of specific diseases and injuries. Distinguishing a medical approach from a health promotion approach should not be taken as a criticism of preventive medicine. The field of health promotion recognizes the need for, and fully supports the incorporation and enhancement of preventive efforts as core IHS services that are integrated throughout the system at all levels of care and across all delivery settings.

Health and Welfare Canada's Strategy for Enhancing Preventive Practice of Health Professionals (1990, 1990a, 1991), defines three levels of preventive medicine. prevention reduces the likelihood of a disease or disorder developing in a person" (Health & Welfare Canada, 1990). Specific examples include periodic health examinations, prenatal care and immunizations. "Secondary prevention interrupts, prevents or minimizes progression of a disease, or irreversible damage from a disease, at an early stage; it comprises the early detection and treatment of disease before irreversible damage has occurred" (Health & Welfare Canada, 1990). Breast and testes self-examinations, early cancer detection, and screening for high blood pressure and cholesterol are examples of secondary prevention. "Tertiary prevention focuses on the progression or damage in a disease where such damage has already occurred irreversibly; the emphasis is on measures to alleviate disability and to slow progression of established diseases or disorders" (Health and Welfare Canada, 1990). Examples of tertiary prevention include monitoring of diabetes, and nutritional and physical activity regimens for those with heart disease. Best practice guidelines for preventive medicine have been developed by The Canadian Task Force on the Periodic Health Examination (1994).

As quality recommendations on incorporating the role of preventive medicine within our health system are readily available from numerous sources, this approach is not developed further in this paper. Instead, we have concentrated on areas where we can make a more significant contribution to the success of IHS reform.

It is our position, however, that preventive medical interventions (like all health services), should be guided by the core values of health promotion. Interactions with clients should foster a sense of empowerment and increase client control over their health, thereby reducing anxiety and dependency. Health education and personal skill development strategies should be employed to enable clients to manage their health to the greatest extent possible. Self-help and mutual support strategies

should also be used where appropriate. Furthermore, practitioners must consider the impact of the social context in which a client lives so as to better meet their health needs.

THE LIFESTYLE/BEHAVIOURAL APPROACH

THE LIFESTYLE/BEHAVIOURAL APPROACH TO PROMOTING HEALTH: AN OVERVIEW

Beginning in the 1950s, epidemiological data revealed the link between multiple causative factors related to lifestyle (e.g. tobacco use, excess alcohol consumption, high-fat diets, lack of physical activity) and the incidence of major non-infectious diseases (Hyndman, 1997). These health risk factors were considered a matter of personal choice, and under the control of individuals. Emphasis was placed on health promotion activities directed towards promoting healthy individual lifestyles and reducing health risks (Tones, 1986; Raeburn and Rootman, 1989). This approach gained support in the 1970s, as a result of growing concern regarding the rising costs of health care. Chronic degenerative diseases had become recognized as a leading cause of morbidity and mortality in the Canadian population, with the bulk of rapidly escalating health expenditures going to expensive medical technologies and facilities directed at treating these conditions (Hyndman, 1997). In an attempt to reduce the incidence of these conditions and contain health care costs, the federal and provincial governments looked to individual/lifestyle health promotion strategies (Labonte, 1993).

The Lalonde report of 1974 (A New Perspective on the Health of Canadians) demonstrated the federal government's official recognition and support of disease prevention and health promotion strategies that addressed lifestyle/behavioural and socioenvironmental risk factors. In particular, the contribution of individual lifestyle risk factors to one's health was emphasized, overshadowing the effects of one's environment. Focusing on personal responsibility for one's health status, the report stated that "individual blame must be accepted by many for the deleterious effect on health of their effective lifestyles" (Lalonde, 1974, 26). This assumption led to the development of multiple health promotion strategies aimed at developing individuals' knowledge, attitudes and behaviours to promote healthy lifestyles and prevent disease.

Strategies included in this approach are largely concerned with individuals or groups whose behavioural or social situation place them at greater risk for developing unhealthy lifestyles. These efforts are supported by a "population approach" used by public health authorities to lower the distribution of a given factor within a larger aggregate, thereby reducing population rates of certain diseases (Labonte, 1997). Targeted behavioural risk factors include: nutrition (diet and weight), physical activity, sexual practices (STDs and unwanted pregnancies), the use of tobacco, alcohol and drugs (prescription and illicit), sun exposure, and injury prevention.

Strategies associated with an individual/lifestyle approach to health promotion include health education, social marketing, brief primary care interventions, self-help and mutual aid, self-care, and public policies aimed at supporting healthy lifestyles.

It is our position that each of these strategies be included as an integral component of IHSs. This position is developed in more detail in Section II.

THE SOCIOENVIRONMENTAL APPROACH TO PROMOTING HEALTH: AN OVERVIEW

While acknowledging the value of interventions that support healthy lifestyles, and advocating for their inclusion within IHSs, we recognize the limitations of a behavioural/lifestyle approach to promote health. By the early 1980s, it was recognized that emphasizing one's personal responsibility for health while neglecting the social and environmental conditions which inhibit the adoption of healthy lifestyles tended to "blame the victim" and ignored issues of social responsibility (Crawford, 1977; Labonte and Penfold, 1981; Buck 1984; Freudenberg, 1985). Furthermore, the approach was proving most effective with the better educated, more privileged members of society. The health status of high risk populations was not being significantly affected, and inequities in health were not being reduced (Labonte, 1993).

Upon critical re-examination, the field of health promotion became more fully aware of the impact that psychosocial and socioenvironmental factors have on one's health. Not only do these factors influence an individual's physiological and behavioural health risks, these conditions are, themselves, independent health risks (Labonte, 1993). This awareness led to the development of a socioenvironmental approach to health promotion.

A socioenvironmental approach to health promotion is founded on the evidence that health is affected by many factors outside the sphere of traditional health care. In this approach, concern is redirected from behavioural risk factors to socioenvironmental risk conditions. Risk conditions refer to socioenvironmental factors that deny individuals or groups of people access to the prerequisites for health. As defined by the World Health Organization, these prerequisites include "peace, shelter, education, food, income, a stable ecosystem, social justice and equity" (1986).

THE SOCIOENVIRONMENTAL APPROACH

THE SOCIOENVIRONMENTAL APPROACH TO PROMOTING HEALTH: AN OVERVIEW CONT.

By the early 1980s, research from multiple fields was beginning to suggest that risk conditions and psychosocial risk factors may be more significant determinants of health than physiological or behavioural risk factors, a claim that appears to be supported by current population health research (Evans et al., 1994). Within the field of social psychology multiple studies demonstrated the importance of social networks and social support on health status (Berkman and Syme, 1979; Israel, 1985; Gottlieb, 1987). Meanwhile, repeated social epidemiological surveys documented the significant correlation between socioeconomic status and inequities in health (Black et al., 1982, 1988). These findings reveal that people living in high risk conditions independently have more disease and premature death and less well-being than average (Labonte, 1997). Over the past decade, population health research has reaffirmed these findings, reporting that psychosocial risk factors and risk conditions are probabilistically associated with individual morbidity and mortality, and population health status (Evans et al., 1994).

Individuals and groups living in high risk conditions often internalize the unfairness of their social circumstances, and blame themselves for their situation. These reactions are reinforced by our culture's emphasis on individual freedom and responsibility. What remains hidden is the degree to which our society's economic and political practices and our dominant ideologies structure our living and working conditions (Labonte, 1997). As a result, self-blame increases feelings of powerlessness and other psychosocial risk factors which are associated with poorer health outcomes. Further research reveals that often people living in high risk conditions adopt unhealthy lifestyles as a way to cope with or bring pleasure to their undesirable situation (Labonte, 1997). For many, the adoption of a healthier lifestyle is of lower priority than the need to address the basics of housing, transportation, food security, etc. (Labonte, 1997). All of these factors serve to increase the prevalence of health threatening physiological risk factors of people living in high risk conditions.

Strategies falling under a socioenvironmental approach to health promotion are directed at reducing and preventing risk conditions, unlike the other two approaches that target high risk individuals or groups. Examples of socioenvironmental determinants of health targeted by this approach include: income, food security, pollution, shelter, employment and working conditions, education, social support, violence, and legal issues.

Strategies associated with a socioenvironmental approach to health promotion are strengthening community development and community economic development, creating environments supportive of health, and advocating for and developing healthy public policy.

It is our position that these strategies be included as integral components of IHSs. Furthermore, to effectively influence the impact of these broader determinants of health, part of the IHS mandate should be to work collaboratively with the communities they serve, other IHSs, the Ministry of Health, and other sectors. This position is developed in more detail in Section III.

IHS ACCOUNTABILITY

The Ministry of Health has stated its commitment to improve the responsiveness, effectiveness, efficiency and accountability of the health system, intending that IHSs be outcomes oriented and evidence based. In order for the Ministry to appropriately set standards against which IHSs will be evaluated, a comprehensive set of indicators must be identified and used to measure health outcomes. These indicators need to reflect the position that health is more than the absence of disease; that it is a positive concept that includes physical, mental, social and spiritual dimensions. Holding IHSs accountable to a comprehensive set of health status evaluation indicators is the strongest, most effective way for the Ministry of Health to ensure that our health system promotes health in addition to treating sickness.

Clearly, neither the health system nor individual IHSs can be held solely accountable for the health status of the population served, or for the risk conditions and psychosocial risk factors that affect the health of Ontarians. Due to the broader determinants of health, this needs to be a collective responsibility that is shared by all sectors, public and private, and all government Ministries. The importance of socioenvironmental determinants of health needs to be better understood and supported by deliberate, collaborative actions and policies. Despite this acknowledgment, it must be recognized that the fundamental purpose of the health system is to promote the health of Ontarians. Therefore, IHSs should be held accountable for incorporating the three approaches to promoting health into their core business. IHSs and the Ministry of Health should also be held accountable for working collaboratively with other agencies and sectors on initiatives that effectively target the broader determinants of health. The outcomes of these efforts should be evaluated by the Ministry of Health.

Within the Ministry of Health's IHS evaluation framework, morbidity and mortality rates are appropriate indicators of the effectiveness of a medical approach to promote health status. Particular attention should be given to measurements related to illness, disability and premature death from specific and/or prevalent diseases, such as heart disease, stroke, cancer, mental illness and AIDS/HIV infection. Other examples of appropriate health status evaluation indicators include: the adoption of best practices regarding screening and immunization coverage; reduced perinatal and infant mortality and long-term morbidity of perinatal origin, reduced disability among physically and mentally impaired citizens, and reductions in cholesterol levels/high blood pressure.

The Ministry of Health should also incorporate evaluation indicators that reflect the effectiveness of individual/lifestyle health promotion strategies and their impact on the health of the populations served. Examples of appropriate indicators include: reductions in consumption of dietary fat; decreased prevalence of overweight; increased rates of physical activity; reductions in the use of tobacco and illicit drugs; reductions in the hazardous use of alcohol and prescription drugs; and reductions in the incidence of STDs, unwanted pregnancies, and injuries related to personal behaviours. A more detailed list of suggested evaluation indicators related to a lifestyle/individual approach to health promotion can be found in Appendix A.

Supporting the position that health is a positive resource for everyday life, indicators that are based on assets, or positive attributes at an individual and/or community level need to be included in the Ministry's evaluation framework (Jackson, 1995; Rajkumar, 1997). Furthermore, social indicators appropriate for a socioenvironmental approach to health promotion must incorporate subjective notions of "wellness" and "quality of life," as well as subjective and objective measures of "social justice" and "social well-being" (see Appendix A for suggested quality of life and social well-being indicators). In order to measure subjective concepts such as what we mean by a "well" society, evaluations should include the perspectives of clients or populations served. This approach requires meaningful public debate in the setting of goals, purposes, guiding principles and evaluation indicators for IHSs at both a provincial and local level, and is consonant with health promotion's emphasis on public participation (Eyles, 1994). It also demands that evaluations of IHSs impact on health include qualitative and quantitative measures. Not all areas of concern regarding the promotion of health can be quantifiably measured (e.g. wellness), and for these concepts rigorous qualitative evaluation methods are appropriate, valid and useful (Eyles, 1994).

Finally, evaluation indicators need to reflect health status outcomes at both a population and a local community level. Population level indicators (e.g. epidemiological data, prevalence of risk factors across the province according to sociodemographic characteristics) can provide a profile of population or community health, against which the entire IHS system and individual IHSs can be evaluated. Population-based indicators of health are unlikely to be sensitive enough to reflect the degree to which IHSs effect the health and well-being of the most marginalized or at risk populations (e.g. the homeless, people living with AIDS, refugees, people with mental illness). Therefore, social indicators are needed to evaluate the extent to which an IHS has addressed social inequities and discrimination by narrowing the gap in health status between those most at risk and the general population. By adopting these types of indicators, the Ministry of Health would hold IHSs accountable for meeting the special needs of marginalized groups on their roster, as well as those who may be unenrollable (e.g. homeless, refugees). Local community level indicators should be developed by each IHS according to provincial guidelines, in partnership with the community served. This will enhance the ability of individual IHS to properly assess the unique health needs of their communities, and the effectiveness of their efforts in meeting those needs.

II: AN LIFESTYLE/BEHAVIOURAL APPROACH TO HEALTH PROMOTION: KEY IHS STRATEGIES

A lifestyle/behavioural approach to health promotion is concerned with the development of personal skills and public policies that support healthy choices and reduce individual risk factors. Key strategies encompassed in this approach include health education, health communication, brief interventions, self-help and mutual aid, self-care, and health promoting policies.

The effectiveness of each of these strategies is described, and examples are provided as to how the strategy could be integrated as a core IHS service. Each strategy is appropriate across all levels of care, with certain strategies (e.g. health education, brief interventions) appropriate or use at all health delivery sites. Within all health delivery domains, and particularly for primary health care, these strategies should be supported by multidisciplinary teams of health practitioners. The cost-effectiveness, quality and value of a collaborative, multidisciplinary approach to primary health care is supported by the WHO (1978) and the Advisory Subcommittee on Primary Care of the Joint Provincial Nursing Committee (1997). To reflect the broader concept of health that IHSs would be working to promote, a primary health team might include a physician, RN, RPN, NP or PHN, nutritionist, social worker, occupational therapist, chiropractor and/or a health promoter.

Each strategy is appropriate and effective across all major risk factors, although effectiveness may be increased by using strategies in coordination with each other. Multi-component strategies may be used to target a specific risk factor or multiple risk factors, and may be directed toward a specific population group, or the community at large. Broad, multi-component community-based strategies, such as the Heart Health Initiatives and Focus Communities in Ontario, may achieve the added benefits of strengthening community action and creating health-supportive environments while promoting healthy lifestyles and reducing individual health risks.

To enhance the effectiveness of such strategies, IHSs would often be required to collaborate with the Ministry of Health, other IHSs, and external organizations or sectors which have an impact on the health of the population. For example, a team of individuals from IHSs across the province could work together to achieve consistent, comprehensive health education, health communication or healthy public policy strategies for the province of Ontario. In other cases, local collaborative efforts involving partners outside an IHS should be supported through IHS' staff and other resources, as part of the system's mandate to foster local intersectoral health promotion efforts.

Again, each of the strategies described through out this section must be guided by the core values of health promotion, outlined in Section IV.

HEALTH EDUCATION

Health education can be defined as "any combination of learning experiences designed to facilitate voluntary adoption of behaviours conducive to health" (Green et al, 1980, 7). This is done through increasing people's knowledge about and awareness of the health benefits and risks associated with certain behaviours. Health education also involves the development of understanding and skill so that individuals are able to adopt positive lifestyles, and resist or change behaviours that pose a risk to their health.

A strong body of evaluation evidence has shown that health education strategies can produce statistically significant changes in health related knowledge, attitudes and behaviours across risk factors, population groups, and delivery sites. Studies have further demonstrated that a multi-faceted approach to support health education strategies can contribute to the achievement of positive health outcomes (see Table I).

Table I: Evidence for the Effectiveness of Health Education Strategies

References:
Kok et al., 1997; Glanz et al., 1990
Glanz et al., 1990; Bracht, 1990
 Sheehan et al., 1996; Tobler, 1994; Botvin & Tortu, 1988; Rundall & Bruvold, 1988 Shah, 1994 Strecher et al., 1994
• Contento et al., 1995
• Dishman & Buckworth, 1996; Schooler 1995
 Kirby et al., 1994; Kim et al., 1997; Holtgrave et al., 1995 Hartley & O'Connor, 1996 Ontario Injury Prevention Resource Centre,
1996; Smith Ulione, 1997 Tobler, 1994; Botvin & Tortu, 1988; Contento et al., 1995

Too often, however, health education strategies have focused on changing knowledge and attitudes while neglecting to incorporate skill development and a socio-cultural perspective into their efforts. As changes in knowledge and attitudes do not necessarily

result in changed behaviour, evaluation results for this strategy have been mixed. In order to enhance the effectiveness of health education interventions, comparative evaluation studies have been used to identify best practices and critical success factors for achieving desired behaviour change (see Table II). It is our position that all IHS' health education efforts be developed according to these guidelines to optimize positive health outcomes.

Table II: Best Practices for Health Education Strategies

Best Practices for Increasing the Effectiveness of Health Education Strategies:	References:
Personal skill development components significantly increase programs' effectiveness.	Sheehan et al., 1996; Tobler, 1994; Botvin & Tortu, 1988; Rundall & Bruvold, 1988; Contento et al., 1995; Dishman & Buckworth, 1996; Oldenberg et al., 1995; Main et al, 1994; Kim et al., 1997; Kelly & St. Lawrence, 1990; Holtgrave et al., 1995; Eldridge et al., 1997
Understanding lifestyles and behavioural choices from a socio-cultural perspective, and designing health education strategies accordingly.	Viswesvaran & Schmidt, 1992; Sheehan et al; 1996; Tobler, 1994; Botvin & Tortu, 1988; Rundall & Bruvold, 1988; Contento et al., 1995; Schooler, 1995; Bryan et al., 1996
Planned and systematic application of behavioural and social science theory.	Botvin & Tortu, 1988; Rundall & Bruvold, 1988; Contento et al., 1995; Schooler, 1995; Schaalma et al., 1996; Holtgrave et al., 1995; Hartley & O'Connor, 1996
Appropriate use of rewards, financial incentives, feedback, and reminders.	Rundall & Bruvold, 1988; Contento et al., 1995; Schooler, 1995
Collaborative development of health education interventions with representatives of the target and user groups, interventions targeted to specific audiences, and interventions that are participatory in nature.	Kok et al., 1997; Strecher et al., 1994; Tobler, 1994; Botvin & Tortu, 1988; Contento et al., 1995; Schooler, 1995; Schaalma et al., 1996
Initiatives producing desired behaviour changes reinforced with a combination of health promotion interventions intended to change social norms (e.g. advocacy, social marketing) and support healthy environments (e.g. healthy public policy, community-based programming).	Contento et al., 1995; Osganian et al., 1996; Llytle et al., 1996; Jeffery et al., 1995; Wilson 1991; Schooler, 1995; Ontario Injury Prevention Resource Centre, 1996

HEALTH EDUCATION: A VISION FOR IHS IMPLEMENTATION

As with all health promotion strategies, health education is most effective when used in conjunction with other health promotion strategies. Multiple human services are therefore often involved in integrated intervention efforts. For example, an IHS might initiate a comprehensive health education strategy in order to reduce the hazardous consumption of alcohol and the incidence of alcohol related motor vehicle accidents within their rostered community.

IHS' staff might work with the education sector to provide a progressive health education curriculum along with general life skills and refusal behaviours in the schools. Some of the system's efforts might go toward developing linkages with the private sector to implement education and training programs for people who sell or serve alcohol. Other staff might work with media and voluntary agencies to develop community educational and social marketing activities such as Drug Awareness Week and campaigns by Mothers Against Drunk Driving. Other educational material, such as posters, pamphlets and self-help resources could be made available in different languages across IHS' delivery sites. The IHS might help to facilitate police involvement in enforcement activities under the Ontario Liquor Licence Act including RIDE (Reduce Impaired Driving Everywhere) and Sober Driver programs. Meeting rooms, support staff and some materials could be provided to assist with the mobilization of the IHS' community in encouraging municipal governments to establish alcohol control policies (Ontario Ministry of Health, 1994). These supports might also be directed towards mutual support groups like AA. Staff may develop programs like Ontario's Focus Communities, which combine adult education, community mobilization to improve high risk conditions, and build social support networks at the same time as they change awareness, knowledge and problem behaviours associated with alcohol and drug use. Primary health teams could use risk factor questionnaires with their rostered members at periodic health examinations to assess for (potentially) problem drinking behaviours, and use brief interventions to change or prevent these behaviours. Furthermore, specific health education initiatives could be developed to integrate with health and health-related social services within the IHS which are directed towards target populations such as children and youth, pregnant women, people who have experienced abuse, and/or the homeless.

A further example of how IHSs could develop comprehensive community initiatives that integrate health education strategies with other complementary health promotion and preventive medicine strategies is provided in the following discussion of Health Communication.

Based on the evaluation evidence presented for the effectiveness of health education strategies to positively impact behaviour change across different health risk factors, target groups, and delivery settings, and the feasibility of integrating health education within IHSs, it is our position that health education strategies should be incorporated as a core IHS service.

HEALTH COMMUNICATION

Health communication incorporates a broad range of health promotion interventions including media advocacy, risk communication, social marketing and other activities focused on individual and environmental determinants of health (Rootman & Hershfield, 1994; Maibach & Holtgrave, 1995). It is a comprehensive approach to health communication involving activities that communicate to, educate, and persuade audiences

so as to promote health. Communications are not limited to lifestyle behaviours, but address any issue which supports individual health and social well-being, such as public participation in civil society (Hershfield, 1997).

Traditionally, social marketing has been the most frequently used approach to health communication. Social marketing can be defined as "the design, implementation and control of programs seeking to increase the acceptability of a social idea or practice in a target group(s)" (Kotler, 1982). Social marketing campaigns involve "the deliberate and intentional use of marketing tools and techniques to plan, implement and evaluate efforts designed to increase support for a wide range of ideas and practices that change the world (at the individual, organizational, and collective levels), for the better (acting in alignment with commonly held health promotion values)" (Hershfield, 1997). Typically, social marketing strategies have attempted to increase audiences' knowledge and awareness of the benefits and risks of lifestyle choices, to promote healthy social attitudes, and to encourage the adoption of health-promoting skills and practices through an organized set of communication activities (e.g. TV/radio public service announcements, posters, booklets, brochures) (Ling et al, 1992; Maibach & Holtgrave, 1995; Freimuth & Taylor, 1995).

Social marketing strategies are distinguished from other health communications strategies by their grounding in corporate advertising and consumer psychology theory. These strategies focus on the beliefs, preferences, needs and characteristics of target audiences, and the "saleable" qualities of the desired outcome or product (Kotler & Zaltman, 1975). Key social marketing concepts include processes of audience analysis and segmentation to better define the group of intended recipients, and the integration of basic marketing elements (price, product, place, promotion and, more recently, participation) to enhance campaign success (Lefebvre and Flora, 1988; Mintz, 1989).

A variety of health communication strategies have been implemented by a wide range of health organizations, including government agencies, local health agencies, and non-profit groups (Ling et al., 1992; Maibach & Holtgrave, 1995). The potential effectiveness of these strategies has been demonstrated in numerous evaluation studies and meta-analyses, for a diverse range of risk factors (see Table 3).

Interventions using media alone are generally most effective in changing awareness, knowledge and stimulating information seeking. These outcomes begin to shift individuals and communities closer to a stage of readiness for actual behaviour change, and contribute to the framing the public agenda around key health issues. In some cases, highly targeted, audience-centred social marketing or health communication campaigns have used a carefully focused message to achieve changes in actual behaviour (Contento et al., 1995). The Stanford Five community-based cardiovascular disease prevention program is a classic example of the potential effectiveness of intensive mass media campaigns to achieve significant changes in health behaviours (Shea & Basch, 1990). The study was based on a community trial in which a control city showed increased prevalence of CVD risk factors during the two year study period, while the experimental community, exposed

to intensive mass media campaigns, witnessed a reduction in risk behaviours (Shea & Basch, 1990). In most cases, however, social marketing and health communications strategies are used in conjunction with other health promotion strategies, in order to achieve significant changes in health-related attitudes and behaviours (see Table 3).

TABLE 3: EVIDENCE FOR THE EFFECTIVENESS OF SOCIAL MARKETING/ HEALTH COMMUNICATION STRATEGIES

Outcome Evaluation:	References:
Significant change in knowledge, attitudes and/or behaviour demonstrated for strategies targeting: • multiple risk factors related to cardio-vascular disease • nutrition • physical activity • HIV/AIDS prevention Significant changes in attitudes, contemplation of and	 Shea & Basch, 1990; Noack & McQueen, 1988 Contento et al., 1995 Schooler, 1995 Wagman, 1993
readiness for behaviour change, and framing the public agenda around key health issues	,
 Results from a meta-analysis of the empirical evidence on effectiveness of mass mediated health campaigns in the United States since 1980: 100% of evaluations found effects on awareness and knowledge, and stimulated information seeking (effects particularly strong when exposure guaranteed; campaigns that saturated a targeted community without guaranteed exposure approximately 25% effective). 85% + evaluations found significant improvements in attitude (guaranteed exposure resulted in considerable changes; general results were more modest). 3 out of 7 studies evaluating behavioral intentions found change rates from 18-73%. 20 out of 29 studies evaluating behaviour change found change rates from 4-74% (median of 29%) 	Freimuth & Taylor, 1995
When combined with other health promotion strategies, (e.g. health education, healthy public policy or community organization), has led to significant changes in health-related attitudes and behaviours.	Farquhar et al, 1977; Flay and Burton, 1990; Redman, Spencer and Sanson-Fisher, 1990; Ling et al, 1992

Ontario's Heart Health Action Program (HHAP) provides an excellent example of how IHSs could use social marketing and health communications as an important component of comprehensive, community-based programs that are developed to enhance the health outcomes of an entire rostered community. Building on similar European and American community initiatives (see Table 4), five Heart Health demonstration sites were developed to learn about effective multiple risk factor community-based programs. Their goal was to reduce risk factors associated with high morbidity and mortality from CVD, and to promote heart health (RBJ Health Management Associates, 1995). Rather than focusing

on high-risk individuals, the HHAP sites targeted various groups throughout their communities (e.g. youth, employees, community leaders and health providers), using a mix of health promotion strategies (e.g. social marketing and health communications, health education, healthy policy, community organizing). Multiple risk behaviours were addressed (physical activity, nutrition, tobacco use), and many sectors were involved in an attempt to build community commitment to the project's overarching goal.

TABLE 4: COMPREHENSIVE, COMMUNITY-BASED BEHAVIOURAL/LIFESTYLE HEALTH PROMOTION INITIATIVES

Previous Community-based Initiatives to References for Program Overviews: - Stanford Three-Community and Five-City Projects - Minnesota Heart Health Program - Pawtucket Heart Health Program - Pennsylvania County Health Improvement Project - Heart Beat Wales - North Karelia Project

HEALTH COMMUNICATION: A VISION FOR IHS IMPLEMENTATION

Similar comprehensive, community-based health promotion initiatives are feasible, and make sense for IHSs. As most people have some level of risk related to preventable behaviour/lifestyle factors, IHSs would be in an excellent position to develop initiatives intended to reduce risk and improve health outcomes across their entire rostered community. IHSs would also be able to ensure that complementary messages are delivered through multiple channels (e.g. school programs, worksite/adult education programs, mass media campaigns, health provider interventions), by facilitating and integrating the active participation of multiple sectors stakeholders within the community.

Clearly, such efforts would reflect a long-term process of change. During the early stages of change, awareness raising strategies are particularly effective at building a profile for the initiative as an important community issue (RBJ Health Management Associates, 1995). The Heart Health sites used specific social marketing/health communication interventions such as a television series on heart health topics, transit ads, regular newspaper columns and ads, and illuminated signs in local public arenas throughout their programs, but particularly during the early stages. IHSs could use similar health communication activities early on in their broad-based community initiatives to increase awareness around specific health-related behaviours and to increase individuals' readiness to make changes in their lifestyle.

IHSs could also use health communication strategies to support and reinforce other health promoting activities. For example, routine personal health risk assessments could be conducted within IHS' primary health delivery sites. Based on the assessments, members

of the health team could provide health education and brief interventions as needed to promote healthy eating and physical exercise amongst clients, and to encourage smoking cessation. The opportunity for similar interventions might also arise with clients admitted to secondary care institutions who are found to have an elevated risk in at least one of these areas. IHS' staff could develop heart health promotion projects for workplaces, schools, and summer camps, and provide training for local personnel and teachers so that they are able to sustain the programs themselves. Supermarket tours, dining guides, and classes teaching low-cost, nutritional cooking skills could be developed by IHS' staff. Educational and self-help resources could be made available, emphasizing a holistic approach to individual and family health, and providing practical tips for action and contacts for support. The IHS could assist local community groups in the development of health-promoting community events, and ensure lots of opportunity for community involvement in related activities and projects. IHS' staff could also assist the community in developing and advocating for healthy public policies such as smoking bylaws, bike lanes, and cafeteria policies.

Given the evidence that social marketing and health communication techniques effectively increase individuals' awareness about the health consequences of their behaviours, and that they have a positive "complementary" effect on other health promotion interventions intended to produce health-related behaviour change, it is our position that these strategies be included as a core IHS service.

BRIEF INTERVENTIONS

Brief interventions optimize the role of health professionals in motivating clients to alter their health risk behaviours (Skinner & Bercovitz, 1996). Clinical encounters between health providers and clients provide an excellent opportunity for brief health promotion and disease prevention interventions. For example, studies show that approximately 10-15% of clients in family practice settings are problem drinkers, and that problem drinkers consult their physicians twice as often as non-problem drinkers (Shah, 1994). Health providers are therefore in a good position to identify clients who are/are at risk of becoming problem drinkers, and offer early brief interventions to reduce the risk of related medical and psychosocial consequences.

Table 5: Evidence for the Effectiveness of Brief Intervention Strategies

Brief Intervention Outcome Evaluations:	References:
Smoking Cessation: cessation rates increased by physicians providing advice during a single routine consultation, or brief smoking cessation messages reinforced on multiple occasions	• Skinner & Bercovitz, 1996; Kottke et al. 1988; Wilson et al., 1988

- tremendous cost effectiveness of cumulative impact of even modest cessation rates (5.8% 10%)
- success rates increasingly effective when interventions target special risk groups (e.g. pregnant women, those with ischemic heart disease)
- extreme cost effectiveness of brief counseling during routine office visits combined with nicotine replacement therapies (e.g. nicotine transdermal patches) to increase cessation rates
- increased effectiveness of advice when used in combination with low to high intensity counseling, more frequent contacts, self-help booklets, and/or the use of nicotine replacement therapies
- Skinner & Bercovitz, 1996; Ockene, 1987
- Dolan-Mullen et al. 1994
- Wasley et al., 1997; Fiore et al., 1994; Silagey et al., 1994
- Ockene, 1987

Problem Drinking:

- effectively reduced/changed alcohol consumption behaviour and achieved treatment referral
- single sessions of sympathetic, constructive advice concerning behavioural self-control strategies, substitutes for drinking behaviour, and brief monitoring consistently found to be more effective than no interventions, and often just as effective as more extensive treatments, in moderating consumption
- Bien et al., 1993; Shah, 1994; Wilk et al., 1997
- Bien et al., 1993; Wilk et al., 1997

Physical Activity:

Increased levels of physical activity in high to low risk individuals through:

- · use of counseling
- video tapes prior to/after health visits
- distributing materials promoting physical activity during routine health visits
- evaluating clients' activity levels using Physician-Based Assessment and Counseling for Exercise model (PACE)
- Lewis et al., 1991; Schooler, 1995;
- Nader et al., 1989
- Campbell et al., 1994
- Calfas et al., 1996

Injury Prevention:

90% of brief childhood injury prevention counseling initiatives in primary care settings reviewed achieved positive outcome effects (e.g. increased knowledge, improved behaviour, decreased injury occurrence)

Bass et al., 1993

Recognizing the potential effectiveness of these opportunities, comprehensive guidelines for preventive counseling services have been developed (Skinner & Bercovitz, 1996). Areas of focus include: diet/weight; physical activity; sexual practices/prevention of unwanted pregnancies; tobacco, alcohol and drug use; injury prevention; and more preventive medical focuses (e.g. breast and testes self-exam). Evidence supporting the effectiveness of brief interventions in achieving desired behaviour change is summarized in Table 5.

These findings and others form an impressive body of research supporting the effectiveness of brief interventions across a variety of risk factors. In particular, autonomy supportive approaches that empower clients, that increase their feelings of self-efficacy around healthy behaviour change, and that motivate them to change unhealthy behaviours have been found to increase the effectiveness of brief interventions across health risk factors (Williams et al., 1996; Botelho & Skinner, 1995). Yet, despite the evidence, and the fact that a majority of adults visit health practitioners annually, expecting that their

practitioner will ask about and provide assistance for health behaviour concerns, practitioners are not routinely raising health risk behaviours with clients (Skinner, 1993; Skinner & Bercovitz, 1996; Wallace & Haines, 1984). Opportunities to maximize brief interventions are avoided due to practitioners' general pessimism regarding their ability to intervene effectively (Skinner & Bercovitz, 1996).

BRIEF INTERVENTIONS: A VISION FOR IHS IMPLEMENTATION

To remove this barrier, IHSs should take advantage of the numerous strategies developed to increase practitioners self-efficacy and increase the likelihood and effectiveness of brief interventions. Educational programs and materials for practitioners that demonstrate the success of skills and approaches to facilitate health-promoting client behaviours have been successful in increasing practitioners use of this strategy (Ockene, 1987). For example, The College of Family Physicians of Canada has developed an Alcohol Risk Assessment and Intervention (ARAI) Resource Manual for Family Physicians (1994) to help educate practitioners regarding this type of brief intervention.

IHSs should also support the use of teamwork, flexible routines and joint problem solving amongst physicians, staff and clients, factors which have been found to enhance and facilitate the use of preventive care interventions in clinical settings (Dietrich et al., 1994; Jaen et al., 1994; Ockene, 1987). IHSs could incorporate the use of a general lifestyle or health-risk assessment survey (self-administered or directed by a member of the primary health team) into routine primary care visits. It should be noted that the use of brief interventions by health practitioners is not limited to physicians but is intended for use by a variety of health practitioners (Botelho & Skinner, 1995). Therefore, based on survey findings, the most appropriate member of the primary health team (e.g. nurse practitioner, nurse, allied health professional, nutritionist, or other lay health worker) could address areas of concern with the client to help them change their behaviour.

Brief interventions are most effectively employed in the primary care setting, as this is usually the first, and therefore most upstream contact that clients have with the health system. However, they are also an appropriate and effective strategy in all IHS' delivery settings. For example, hospital staff could easily incorporate brief smoking cessation interventions into their pre and/or post-operative consultation with clients. The initial intervention could be noted on the client's integrated health file, making it easier for a member of their primary health team to follow up and/or monitor the intended change in behaviour. Staff in long-term care facilities see their clients on a daily basis, and therefore have the ongoing opportunity to identify high risk behaviours and intervene before problems develop. Home care health providers could use brief interventions as needed, for example in the area of injury prevention or smoking cessation during post-natal home visits, or to promote proper nutrition and appropriate levels of physical activity amongst the elderly living at home. By training staff across delivery sites in the appropriate, opportunistic use of brief interventions, IHSs could maximize their opportunity to positively effect health-related behaviours, and enhance the health status of their clients.

Based on the evidence of outcome effectiveness, it is our position that the use of brief interventions be expanded and included as a core service through out IHSs.

SELF-HELP AND MUTUAL AID

Self-help or mutual aid have been described as "a process wherein people who share common experiences, situations or problems can offer each other a unique perspective that is not available for those who have not shared these experiences" (Self-help Resource Centre of Greater Toronto, 1996). Self-help and mutual aid groups are run by and for their members. Any involvement by professionals is limited to a consultative or ancillary role, and is at the request of the members (Kurtz, 1990). Groups are open to all members of the general public, offering free and voluntary help (Kurtz, 1990).

The primary objectives or beneficial outcomes of a self-help or mutual-aid strategy include the exchange of emotional support, self-disclosure, problem clarification, information sharing, friendship, identity formation, personal growth and transformation, advocacy, and collective empowerment (Kurtz, 1990). A review of research literature reveals that self-help initiatives enable people to cope with range of health-related problems (e.g. addictions, bereavement, abuse, cancer, disabilities and mental health issues) (Rogers, 1989; Kurtz, 1990; Medvene, 1990; Powell, 1994; Hyndman, 1996). They also have been found to encourage the adoption of preventive practices amongst members, such as decreasing high-risk injection practices and promoting HIV prevention measures amongst intravenous drug users (Sibthorpe et al., 1994).

Few formal outcome or impact evaluations of self-help and mutual aid groups have been conducted, for a number of reasons. Some barriers to formal outcome evaluations include concerns regarding anonymity and privacy, altering the natural helping process through intrusive evaluation methods, a lack of interest on the part of members to measure impacts and keep files or records, variable group participation, self-selected heterogeneous populations, and the lack of proven causal relationships between positive benefits of self-help or mutual aid groups and changes in standardized, clinical health status outcomes (Kurtz, 1990a; Hyndman, 1996; Romeder et al., 1990; McCrady & Miller, 1993). As a result, qualitative studies that capture the "lived," personal and subjective experiences of group members and the underlying socio-cultural context are a valid and appropriate alternative method for evaluation. Qualitative studies can be used to gather evidence as to the nature and extent of the strategy's impact on members, and explain why or how that impact occurred (Hyndman, 1996).

Despite the challenges to conducting impact evaluations on self-help/mutual aid groups, research has shown this strategy to be effective at promoting the health and well-being of group members. For instance, changes in US cirrhosis mortality rates have been positively associated with changes in per capita consumption and negatively associated with membership to Alcoholics Anonymous (Mann et al., 1991). While a causal interpretation is not possible, the hypothesis of its existence (i.e. lowered levels of per

capita consumption and increased AA membership reduce the drinking levels of individuals who might otherwise die from liver cirrhosis) is supported by Swedish and Canadian data (Mann et al., 1991). Furthermore, members of GROW, a twelve step self-help group designed to help members suffering from a history of psychiatric problems in preventing or recovering from mental illness, were found to have fewer hospitalized days over a thirty-two month period than did a comparable group of non-members (groups were matched on twelve variables including history of hospitalization) (Kennedy, 1989).

Self-help and mutual aid groups are increasingly seen as effective strategies used to support caregivers (Toseland et al., 1990; Toseland et al., 1989) and, more commonly, individuals with disease-related health concerns (Trojan, 1989). A review of the benefits of these strategies as reported by members from 65 disease-related groups found that for the vast majority, participation positively impacted disease-related stress, relationships with family and friends, and patient behaviour and professional services (Trojan, 1989). These findings are supported by an evaluation of Edmonton's Cross Cancer Institute (CCI). The majority of women who attended CCI self-help group sessions for four months described the strategy as a good way to find effective coping skills to get them through the cancer treatment, and felt that the sessions increased their self-esteem (Look Good, Feel Better, 1997). Supportive data have also led the Canadian Cancer Society to develop a Reach to Recovery program in which volunteers who have had breast cancer visit one-to-one with women newly diagnosed with the condition (an impact evaluation is currently underway).

Studies have found that self-help and mutual aid strategies are more likely to achieve positive outcomes if they increase members' degree of participation in the group. Increased participation has been linked to increased satisfaction and self-esteem, reduced treatment utilization, greater coping skills, and more positive attitudes towards health-related problems (Kurtz, 1990; Reissman & Carroll, 1995).

SELF-HELP/MUTUAL AID: A VISION FOR IHS IMPLEMENTATION

IHSs would be in an excellent position to support interested members of their rostered community in developing self-help/mutual aid groups. IHS' staff could provide professional assistance (when requested), meeting space, and other appropriate resources to support these groups. This form of IHS' support should be made known to clients through publicized posters, pamphlets and other modes of communication. IHSs could be particularly helpful in assisting established self-help/mutual aid groups overcome barriers to regular attendance and participation by interested members of the rostered community. IHSs could help to market the groups, their meeting places, and their meeting schedules. They could also provide self-help groups with information, contact numbers, supportive research, and practical tips on the particular issue of interest, increasing participation within groups, ensuring accessibility, and other related matters.

Therefore, it is our position that, as a mandatory core service, IHSs should support the creation and existence of self-help and mutual aid groups by interested members of their rostered community.

SELF-CARE (SELF-GUIDED HEALTH EDUCATION)

Self-care has been described as a set of unorganized health activities and health-related decision making by lay people to promote healthy personal skills and behaviours on their own or their families behalf (Hatch & Kickbusch, 1983). Self-care actions may concern medical problems (e.g. prevent disease, evaluate symptoms, restore health), or they may be aimed at maintaining and improving health (Health Canada, 1997). They have been characterized by several health care disciplines as: "being situation and culture specific; involving the capacity to act and make choices; being influenced by knowledge, skills, values, motivation, locus of control and efficacy; and focused on aspects of health care under individual control (as opposed to social policy or legislation)" (Health Canada, 1997).

Increasing evidence indicates that a substantial percentage of individuals who resolve their health risk behaviours do so without relying on formal or professional help (Sobell et al., 1993; Cunningham et al, 1992; Stall and Beirnacki, 1986). Approximately 80-90% of cigarette smokers quit (Fiore et al., 1990; Orleans et al , 1991), 60-75% of problem drinkers abstain or achieve moderation (Cunningham et al, 1992), and 60% of overweight individuals lose weight (Schachter, 1982) on their own. These rates suggest the need for greater support of self-care or self-guided strategies.

While conclusive evidence on the health impacts of self-care is lacking, a review of the effectiveness of self-care interventions reveals that (1) self-care is widely practiced, (2) self-care actions are often beneficial and seldom harmful, and (3) self-care appears to be a universal behaviour (Health Canada, 1997). Self-care strategies may include planned self-care education programs, booklets, manuals, video cassettes and other tools that are directed to individuals who want to guide themselves to healthy lifestyles or behaviours. Self-care materials most often provide health education related to smoking cessation, moderating alcohol consumption, dietary compliance, physical activity, self-medication, self-treatment, and social support activities that take place within context of people's normal, everyday lives (Hyndman, 1996; Hatch & Kickbusch, 1983).

For example, self-care manuals used in minimal contact smoking cessation programs have been found to be useful in enhancing self-efficacy and self-management skills, key success factors required for permanent smoking cessation (Pederson et al., 1981; Utz et al., 1994). Initial assessments of self-care books offered to persons wanting to quit or cut down on their alcohol use have also been shown to achieve significant short-term effects in moderating drinking behaviours (Sanchez-Craig et al., 1996). Dietary self-care materials promoting healthy eating patterns in the general population suggest the potential for modest positive changes in behaviour, particularly in the preparation of food for families

(Beresford et al., 1992). Similarly, a review of physical activity interventions suggests that providing individuals with information promoting non-structured physical activities that can be done at home at the individual's convenience may be an effective strategy, especially if used in combination with programs delivered in multiple settings (King et al., 1991).

There is also evidence to indicate that use of self-care resources is a cost-effective strategy for encouraging health promoting practices (Hyndman, 1997). Although not all outcome evaluations have found self-care interventions to achieve the sustainability or high rates of behaviour change as more intensive programs, self-care strategies are generally more portable, and therefore have widespread potential for dissemination and broad-based impact on population health (Hyndman, 1997).

SELF-CARE: A VISION FOR IHS IMPLEMENTATION

While the limits to self-care must be acknowledged, this strategy can add significantly to a person's competence and skills (Health Canada, 1997). The aim of self-care should be "to develop a more efficient health care system, one based on fuller exploitation of every component and every competence, and on the dynamic and integral involvement of people" (Perreault & Malo, 1989). IHSs could support self-care strategies in a number of ways.

A precondition for the development of self-care by individuals has been found to be the presence of receptive providers (Barofsky, 1978). Therefore, primary health teams within IHSs need to develop open, autonomy-supportive, collaborative partnerships with their clients. IHSs need to ensure providers have time for consultation and the appropriate transfer of some aspects of care (e.g. assessment, monitoring and treatment) to their clients during client visits. Building clients' self-efficacy (the belief that one is capable of dealing with a specific problem) and self-concept have been reported to play a potentially greater role in self-care than having information or skills (Health Canada, 1997). Therefore, health providers could support clients through counseling that addresses these factors, and client motivation.

Education and intervention strategies could be developed to increase clients' knowledge and skills, enabling them to more actively participate in creating their own health. Providers could discuss with clients the activities they envision themselves engaging in to promote their own health. Reinforcing positive health behaviours with feedback and follow-up, assisting clients to develop short- and medium-term goals for behaviour change, and helping clients build support from family, social networks and their community are all ways IHS staff could support self-care. Workshops that focus on self-care could be organized for clients in various settings, and user-friendly self-help materials related to a range of lifestyle issues could be developed to address the health and language needs of different client groups. These tools could be placed in all IHS' primary health delivery sites near to the comprehensive lifestyle or risk-assessment survey previously mentioned. Clients would therefore be able to select material related to areas of personal concern.

IHSs could make self-care materials available at community health delivery sites, long-term care facilities, pharmacies, community health resource centres, worksites, and schools. IHSs could integrate this strategy into their primary health practice with a well-developed assessment of clients' self-care behaviours (a framework to support this approach has been developed by Health Canada), and by recommending self-care tools, materials and programs to clients (Health Canada, 1997). Finally, self-care strategies could become an integral part of other IHS' health education and brief intervention activities.

Based on initial evaluation data it is our position that, as a core service, IHSs provide support materials to assist clients attempting to promote their own health through self-care strategies.

HEALTHY PUBLIC POLICY

The Ottawa Charter for Health Promotion defines the building of health promoting public policy as a combination of "diverse but complementary approaches including legislation, fiscal measures, taxation and organizational change" (WHO, 1986). This strategy demands coordinated action and shared responsibility amongst policy makers in all sectors and at all levels, requiring that they "be aware of the health consequences of their decisions and to accept their responsibilities for health" (WHO, 1986). The aim of healthy public policy in relation to an individual or lifestyle approach to health promotion is to prevent unhealthy public risks and to facilitate and support the adoption of healthy choices.

Policy measures to control tobacco generally have widespread support within the health system, and their effectiveness is backed up by strong scientific evidence. Policies on pricing have been shown to have a strong impact on consumption of tobacco products. With every 10% price increase, studies from Canada, Europe, the United States and New Zealand have found a corresponding drop in consumption of 3-6% (Godfey & Maynard, 1988; Townsend, 1988; Pekurinen, 1991; Sweanor, 1991, 1992; Andrews & Franke, 1991; European Union/HEA, 1995). Tobacco taxation policies have the greatest impact on adolescents and low-income groups, whose consumption patterns are more price-sensitive (Sweanor, 1991; Townsend et al., 1994). While effective, pricing policy strategies are most difficult for low-income smokers, whose disposable income may be significantly reduced as a result (Marsh & McKay, 1994). Therefore, pricing policies should be used in combination with policies targeting availability, distribution and promotion, and with interventions that assist these populations in smoking cessation and in enhancing their living conditions.

Pricing policies are also one of the most effective policy strategies for reducing the consumption of alcohol, and for reducing the harmful social consequences of dangerous alcohol consumption, such as motor vehicle accidents and deaths (Moskowitz, 1989; Edwards et al., 1994). Although to a lesser extent than with tobacco, taxation rates for

alcohol effect the consumption patterns and the disposable incomes of low-income individuals the hardest. To be most effective, pricing policies should be used in combination with other policies around the system of alcohol distribution, minimum drinking age laws, restrictions on advertising, municipal alcohol legislation, and legislation requiring liquor licence holders to provide their staff with server training interventions, and with other health promotion strategies (Ontario Task Force on the Primary Prevention of Cancer, 1995).

Nutrition policies are internationally recognized as an appropriate strategy for the promotion of healthy eating and the reduction of diet-related chronic diseases (Edwards, 1996). Dietary policies include the development and promotion of dietary guidelines such as those found in the Canada Food Guide, and economic strategies to ensure access to a high-quality, affordable supply of nutritious food (e.g. reducing the price of whole wheat flour, lowering the duty on imported fruits) (Posner et al., 1994). Other examples of suggested policy strategies include an intersectoral approach between government, the agri-food industry, food retailers and food services to ensure the comprehensive implementation, monitoring and evaluation of policies to promote healthy eating habits in stores, restaurants, and school/worksite/health organization cafeterias, and the development of an adequate food labeling system to assist consumers in making informed nutritious food choices (Ontario Task Force on the Primary Prevention of Cancer, 1995).

In regards to physical activity, this type of "passive" intervention strategy is often found to be more successful at achieving population wide changes than those requiring active decision making by individuals (King, 1995). From posting point-of-choice messages next to stairwell entries, elevators and escalators in public buildings, which significantly increase stair use by both obese and fit people, to multi-policy community-based strategies, physical activity can be positively effected (Brownell et al., 1980). For example, a community-based intervention combined legislation with other health strategies to significantly increase their community's fitness levels, including levels among persons with substandard fitness levels at baseline (Linenger et al., 1991). The intervention included increasing access to recreational facilities and extended their hours, building bicycle paths, installing new exercise equipment in gyms, scheduling community-wide athletic events, opening women's fitness centres, marking running courses throughout the community, organizing running and cycling clubs, encouraging employees to provide release time for physical activity, and initiating rewards for improved physical performance. In another low-income community physical activity levels were moderately increased through policies that upgraded community recreation facilities and enhanced general safety precautions (i.e. the presence of police during outdoor exercise session), thereby making activity more convenient and safe (Lasco et al., 1989).

Various health bodies have recommended the development and implementation of appropriate policies targeting other diverse unhealthy public risk factors, including sun exposure (e.g. mandatory available shaded areas at all school yards, beaches, playgrounds and other outdoor public and work places where feasible, and a labeling system to identify the degree of UV protection for all articles of recreational clothing) (Ontario Task Force

on the Primary Prevention of Cancer, 1995), healthy sexual practices (e.g. availability of condom machines in school and public restrooms, subsidized pricing for contraceptives), and injury prevention (e.g. road safety standards, speed limits, the creation of bicycle lanes, mandatory bicycle helmets).

HEALTHY PUBLIC POLICY: A VISION FOR IHS IMPLEMENTATION

IHSs could allocate a proportion of financial resources and staff time to develop, research and advocate for the implementation of healthy public policies that promote and support healthy lifestyle choices. Healthy public policy efforts can be directed towards multiple levels of government, in multiple sectors. In order to improve the outcome effectiveness of this strategy, IHSs could work with other IHSs, as well as initiate intersectoral collaborative efforts towards a common cause. IHSs could also support the mobilization of members of their rostered communities to advocate for healthy public policy around issues that most concern them. IHS' support could include the provision of meeting rooms, research support, administrative support in drafting reports or position papers, and staff support in organizing and outreach efforts.

It is our position that a core IHS service should involve development, research and advocacy efforts to promote the implementation of healthy public policies. As part of this core strategy, IHSs should be required to demonstrate their involvement in a certain amount of intersectoral collaborative efforts, and their support for community mobilization and advocacy around healthy public policy issues.

III: A SOCIOENVIRONMENTAL APPROACH TO HEALTH PROMOTION: KEY IHS STRATEGIES

A socioenvironmental approach to health promotion is concerned with strengthening community action, building healthy public policy and promoting environments that are supportive of health. Key strategies encompassed in this approach are described below. They include community development and community economic development programs, healthy public policy and health advocacy, and developing comprehensive health promotion initiatives for schools, workplaces and communities.

The contribution each of these strategies can play in addressing the health impacts of broader socioenvironmental issues is described, and examples are provided as to how the strategy could be integrated within an IHS. It is our position that each of these strategies be incorporated as a core IHS activity area. This does not mean that these strategies should be the sole responsibility of IHSs. Clearly, they will require a collaborative effort with partners outside the IHS. However, IHSs should be held accountable for incorporating these health promotion strategies into their core business. IHSs and the Ministry of Health should also be held accountable for working collaboratively with external health-related agencies and sectors on provincial and local level initiatives that effectively target the broader determinants of health. Collaborative initiatives of this nature should be supported through IHS' staff and other resources as part of the system's mandate to address the impact of the broader determinants of health, and to foster local intersectoral health promotion efforts. Anticipated partners will include grassroots community groups, local schools and/or school boards, municipalities, religious organizations, businesses, social service and related health organizations that may be external to the IHS, other IHSs throughout the province, and various provincial Ministries.

Again, each of the strategies described through out this section must be guided by the core values of health promotion, outlined in Section IV.

COMMUNITY DEVELOPMENT

Community development (CD) has been defined as "the process of supporting community groups in identifying their health issues, planning and acting upon their strategies for social action or social change, and gaining increased self-reliance and decision-making power as a result of their activities" (City of Toronto, 1993). The fundamental purpose of a CD strategy is to improve people's immediate psychosocial health and to strengthen community action, or the collective ability to influence those living and working conditions that effect health. This outcome requires changes in the thinking of community members, and their willingness to participate in planning and implementing actions so as to achieve better health.

A CD strategy first requires clarity about the community to be developed. For IHSs, all rostered members may be considered a community. Yet within that rostered community, there are other groups of people who share a collective identity, and a sense of collective purpose. These groups, or communities, may or may not live in the same local neighbourhood. They may be a community based on a shared belief system, common interests, or some other defining characteristic. Therefore, IHSs, like most neighbourhoods and towns, will have multiple communities within them. A CD strategy may be used to support any type of community group in identifying and acting to resolve the concerns and issues that are important to them (Labonte, 1993). These efforts may be most leveraged, however, when they are used to build community capacity within groups living in high risk conditions, so that they are better able to address the root causes of health and social inequities.

The major underpinning of CD is the notion of empowerment (Wallerstein, 1993; Eisen, 1994; Raeburn, 1996, WHO, 1986). Empowerment has been defined as the capacity for choice. It includes "the ability to define, analyze and act upon problems one experiences in relation to others, and in one's social and environmental living conditions. Empowerment as a process describes the means through which internal feelings of powerlessness...are transformed, and group actions initiated to change the physical and social living conditions that create or reinforce inequalities in power" (Labonte, 1993). Empowerment is what distinguishes CD from many other broad, community-based health promotion strategies, like the heart health initiatives described in Section II.

In community-based health promotion strategies, health professionals and/or health agencies defines a health problem to be addressed (that is usually related to the prevention or reduction of a health risk factor), rather than the community residents. In community-based strategies, health agencies develop specific strategies or interventions to resolve the problem (e.g. health education, brief interventions, healthy public policy), that are implemented according to defined timelines (Hyndman, 1997). While local community members and groups are encouraged to actively participate in community-based health promotion efforts, and, ideally develop the capabilities to take ownership and responsibility for maintaining these efforts, community based strategies are not underpinned by the principle of empowerment (Labonte, 1993; Boutilier, 1996).

In contrast, community development begins with the community defining the issue or problem to be addressed (usually a socioenvironmental factor). Continual negotiation between a community worker, community organizations and community groups enable community participants to critically analyze their broader living conditions, and work together to improve them. The community, not the health authority, is the primary decision-maker with regard to the planning and implementation of health promoting strategies (e.g. mobilization, social action). Furthermore, the outcome focus of CD is enhanced community capacities and improved living conditions, rather than measurable changes in specific health risk factors (Hyndman, 1997).

These distinctions are not a criticism of broad, community-based health promotion strategies. As has been shown in Section II, they are an effective means of reducing health risk behaviors and promoting healthy behaviours within a community, and should therefore become core components of IHSs. The intent of this review is to achieve a common understanding of the process, purpose and value of community development, so that it too can be incorporated as a core IHS area of activity.

Although CD strategies require a long-term impact evaluation, evidence supporting the effectiveness of this approach can be found from a variety of sources. Perhaps one of the most well known projects is the Tenderloin Senior Organizing Project (TSOP), of San Francisco (Minkler & Cox. 1980; Minkler, 1985, 1990, 1992). The project has helped low income seniors meet their health needs for over two decades. Established in the mid 1970s, TSOP responded to the needs of elderly residents of local hotels and rooming houses. Residents were living in extremely high risk conditions, including poverty and social isolation. They had internalized strong feelings of powerlessness, and were plagued by health problems including alcoholism, hypertension, and malnutrition. By slowly gaining the trust of residents, community development workers organized residents to discuss and share their concerns. Gradually, hotel-based coalitions were formed. One of the first issues participants wanted to address was the problem of malnutrition, which was partly the result of a lack of access to fresh fruits and vegetables. Two action plans were developed and implemented. A collective of residents from three hotels arranged contracts with a local food service and established their own hotel-based mini-markets one morning per week. A group of residents in another hotel established a cooperative breakfast program, which qualified their residents for participation in a local food bank. Other initiatives have focused on concerns around crime and violence. They include recruiting local businesses and agencies to serve as "safehouses" where seniors can go in times of emergency. A participant advocacy effort was successful in increasing the number of patrol officers in the neighbourhood, which, along with other measures, brought about an 18% reduction in crime during the year following the project's beginnings (Minkler, 1992). Physical and mental health indicators have improved as the result of stronger social networks, relevant health education, and increased access to the determinants of health (Minkler, 1992). Through continued support for lobbing and advocacy skill development, and leadership and communications training, TSOP has sustained community involvement and strengthened community capacity to successfully promote the health of its members (Minkler, 1992).

Evidence also comes from two winning entrants of the Caring Communities Award sponsored by the Trillium Foundation (Trillium Foundation, 1997). Both communities were recognized for their outstanding leadership in community development initiatives that addressed some of the most pressing issues that are being faced by Ontario communities. In Cornwall, for example, a diverse range of community stakeholders, from residents, local industry, Boy Scouts, and the Kiwanis and Rotary Clubs, worked together on a number of initiatives to protect and enhance their local environment. Their efforts were sparked from community concern about how local economic dependence on textile and paper industries was negatively impact the health of the natural environment, and the community.

Collaborative efforts included the development of an Ecology Park and Community Garden on 4.5 acres of St. Lawrence River Shoreline. The site includes a Community Garden for seniors and low-income residents, a demonstration of how Domtar industrial waste can be used as a fertilizer, a central pond, and a Pavilion.

Successful community-wide development initiatives also enhanced the health of the West Bay First Nation community on Manitoulin Island (Trillium Foundation, 1997). Over the last 15 years, West Bay residents have come together to revive the Native culture that had traditionally bound their community together. Their efforts revitalized a sense of pride and self-identity amongst residents, and generated further energy for community development projects. Community recreation facilities have been built, and the youth of West Bay have become involved in a range of sporting activities. Local jobs and economic growth have been created as a result of a new found entrepreneurial spirit. As the community grows stronger and more vibrant, residents are celebrating the return of educated young people who are now moving back into the community, and there is a growing sense of optimism about the community's capacity to care for the economic and social well-being of its people.

COMMUNITY DEVELOPMENT: A VISION FOR IHS IMPLEMENTATION

Adopting asset based planning like that used by the York Region District Health Council, and by providing resources, training, facilitation and other sources of support to interested community groups, IHSs can enhance the health-promoting capacity of its rostered community. When communities are empowered in this way, they can truly work in partnership with the formal health system towards the shared goal of improved, and equitable health and social well-being.

Based on the value and feasibility of incorporating community development into the ongoing work of an IHS, it is our position that this strategy should also become a core IHS area of activity. A CD strategy will enable IHSs and the communities they serve to really focus upstream, working to address the most fundamental determinants of health and of health and social inequities. By strengthening community capacity, and empowering communities to have greater control over the factors that determine their health, CD has the potential to help IHSs achieve the most sustainable positive impacts on health and social well-being for the people of this province.

COMMUNITY ECONOMIC DEVELOPMENT

Community economic development (CED) is a health promotion strategy that builds on community development. The distinguishing characteristic between the two strategies is CED's focus on addressing the impact of economic factors on communities. Strong evidence points to the importance of adequate income, meaningful employment opportunities, and healthy work environments as fundamental determinants of health

(Wilkinson, 1986; Mustard & Frank, 1991; Evans et al., 1994). Macro-level trends and policies (e.g. long-term unemployment, fiscal restraint policies) can impair equitable access to these determinants, thereby negatively effecting the health of individuals and communities (Hyndman, 1997). Community economic development is a strategy for strengthening communities' economic capacity and self-sufficiency, so that they are less vulnerable to the negative impact of macro-economic trends. The strategy has been defined as "a process by which communities initiate and generate their own solutions to shared economic priorities to enhance their economic and social well-being" (Ross & Ushser, 1986).

An example of a successful CED strategy is the story of Bethel New Life, a community development corporation. New Bethel Life was initiated by two church-based volunteers living in a Chicago neighbourhood. When the local hospital announced plans to close, Bethel New Life mobilized their community and raised the funds needed to buy the hospital's campus. Starting with a budget of \$9,600, the community group now has a \$6.5 million dollar budget. Working with local businesses, community leaders, citizens, government and other sectors, the group developed a day care centre, drug store, bank, and training institute for office workers and certified nursing assistants in the old hospital buildings. It now has plans for converting the main building into housing units for elderly and adult day care, and in the future opening a primary health care centre on the grounds. The community economic development strategy was founded on the principles of empowerment and self-sufficiency. Building on community assets and resources, the group identified that many people in their community had the skill base to become caregivers, yet they lacked the required formal certification. In response, they developed a local training institute for certified nursing assistants housed in a local building site (the old hospital). The collaborative effort was organized to address issues that the community itself identified as a priority, such as the need for services for the elderly, and job training for people in community. Broad-based communication, public involvement, and partnerships were continued throughout the process, with community leaders guiding the strategy and decision-making around the action plan, and health practitioners serving to facilitate and support the process. Not only was the strategy a major economic and employment success, New Bethel Life developed the community's capacity to exert greater control over the factors that effected its health and well-being (Zablocki, 1996).

Local CED strategies can be found in a number of Ontario communities. For example, the Milverton-Mornington Revitalization process was developed by residents of Milverton to address the closure of the Deilcraft furniture factory. The factory was the largest employer in the community, and its closure left many people unemployed. To improve their economic and living conditions, a community group formed with purpose of revitalizing economic opportunities in the community using, as much as possible, the skills of the unemployed workers. Some success was achieved in buffering the effects of the factory closure, and fostering economic self-sufficiency in the community (Bennet, 1992).

Residents in Ear Falls, a finalist for the Trillium Foundation's Caring Communities Award, joined together in response to the devastating economic effects from the mid-1980's

closure of the town's main employer (Trillium Foundation, 1997). Based on an enduring commitment to build a supportive and open community, local efforts have won new investment and jobs. A proactive community spirit and a strong relationship with local First Nations communities were two of the critical factors that convinced Avenor Inc. to invest \$60 million in a new sawmill in Ear Falls.

In Collingwood, another Caring Communities Award finalist, the Healthy Community Council mobilized different stakeholders in the community to rebuild the region's economic base (Trillium Foundation, 1997). Since 1986, Ontario's leading shipyard had entered a period of uncertainty and business decline. Focusing new efforts on tourism, the community has worked together to restore the town's harbour, and transform itself from a once designated "Area of Concern" to the only North American site to be given a clean bill of health by the International Joint Commission. Impressive growth in tourism has been balanced with a sensitivity to local residents, some of who were dislocated. Conscious, innovative efforts have been made to retrain the area's workforce and draw upon community resources in order to meet the demands of the region's new area of economic development.

Different health agencies and providers also support CED strategies on a smaller scale. For instance, the Black Creek Focus Community, a community coalition that takes a CD approach to addressing alcohol and drug related health risks, supported community members in setting up a local bakery. The project was not only intended to generate income, but to develop community members' skills in negotiating and setting up a business, to build on members' existing skill sets (baking, handling money), and to develop a sense of self-efficacy and pride in what community members could collectively accomplish.

COMMUNITY ECONOMIC DEVELOPMENT: A VISION FOR IHS IMPLEMENTATION

IHSs could help members of their rostered community organize and mobilize to address their concerns. Resources such as staff time, planning and administrative skills, and expertise in writing grants and facilitating intersectoral collaboration are all practical ways IHSs could support local efforts. IHSs could also develop CED strategies as ways to assist persons with mental and physical disabilities, and other groups experiencing high risk conditions that negatively impact their access to the basic necessities for health. The Mad Market, a non-profit, used goods store was developed in 1980. Staffed and controlled by mental health consumers, the store provided members of that community with valuable job training, skill development, self-esteem and paid employment (Weitz, 1988). Community groups could explore and develop many more options with the support of IHSs.

Based on the reality that individual and community health is primarily determined by social and economic issues, it is our position that community economic development should become a core IHS area of activity. Enabling IHSs and the communities they serve to really focus upstream and address the most fundamental determinants of health, CED is a key strategy to promote health and to redress health and social inequities. Furthermore, by strengthening community economic capacity and self-sufficiency, CED, like CD, offers the potential for achieving the most sustainable positive changes in health and social well-being.

HEALTHY PUBLIC POLICY and HEALTH ADVOCACY

In Section II, the strategy of healthy public policy was defined as it relates to an individual/lifestyle approach to health promotion. In that context, healthy public policy refers to policies from any sector that are intended to reduce risk behaviours or promote health behaviours.

In relation to a socioenvironmental approach to health promotion, the strategy takes on a different dimension. "There are many policies that have a direct bearing on the extent of inequality in our society, and thus on the extend to which people from different social circumstances have access to health related resources" (Link & Phelan, 1995). Such policies are often considered to lie beyond the realm of responsibility of the health system. Yet, if root causes of health are these broader living and working conditions, then the potential impact of these broad policies needs to be thoroughly understood and addressed by the health system. As a strategy addressing the broader determinants of health, healthy public policy "is characterized by an explicit concern for health and equity in all areas of policy and by an accountability [in each of those areas] for health impact" (The Adelaide Recommendations, 1988). Healthy public policies can be developed to address the multiple risk conditions and prerequisites to health described in Section I. Whichever risk condition they target, however, they are united by a common goal of redressing structures of social inequities, which have been identified as the root causes of health inequities (Edwards, 1996; Evans et al., 1994; WHO, 1992).

The key factors that determine whether a public policy is "healthy" is that it has a positive impact on health (as it is broadly defined), that it is directed towards achieving social equity and justice, and that intersectoral collaboration and public participation were involved in the development and implementation of the policy (Draper, 1988; Edwards, 1996).

Clearly, the best way to enhance community health and well-being is to create healthy communities. Healthy public policies can serve as an effective framework for helping us shift in this direction. Often, however, the general public equates "healthy policies" with "health policies" related to the organization, funding and delivery of health care (OPHA, 1993). Therefore, IHSs and our health system should develop awareness and education campaigns that help communities understand the impact that their living and working conditions have on health. Once people understand the concepts of and evidence relating to the socioenvironmental determinants of health, they will be better able to make informed decisions and advocate for the public policies they desire to support health and

social well-being. In addition to awareness and education efforts, IHSs should allocate a proportion of financial resources and staff time to research and develop healthy public policies that promote and support healthy environments.

France provides an excellent demonstration of the potential effectiveness of a public policy strategy to promote health. In the early 1980s, the French government passed a number of social and economic policies directly aimed at reducing the risk of premature birth (Kushner & Rachlis, 1994). For example, the government paid women to attend prenatal sessions, and provided them with food supplements during pregnancy. These policies helped to redress income inequities, and ensure that all women had equal access to prenatal education, support, and adequate nutrition, factors that are critical to both the health of the baby and mother (Walker, 1991). Maternity leave before delivery was extended to nine weeks, and pregnant women in Paris were given 30 minutes off at the start and end of every working day to help them avoid the most hectic part of rush hour. Through the collaboration and commitment of national and municipal governments, health authorities, labour unions, and private and public worksites who supported this legislation, the rates of prematurity dropped by 30%, with a 50% decrease in the rates of the lowest birthweight babies (Rachlis & Kushner, 1994). France now has one of the lowest infant mortality rates in world, an outcome achieved primarily through healthy public policy. Not only was policy development a cost effective strategy for avoiding ongoing and costly medical interventions, it also prevented a lifetime of disability and health problems often associated with Low BirthWeight babies (Rachlis & Kushner, 1994).

HEALTHY PUBLIC POLICY: A VISION FOR IHS IMPLEMENTATION

Here in Ontario, IHSs could initiate and support the development of similar healthy public policies that meet the needs of women before, during and after pregnancy, as well as addressing the health of children and families. For example, IHSs could work with community projects like "Healthy Beginnings" and "Best Start" to develop policies for providing food supplementation to pregnant women, infants and children. They could collaborate with low-income housing projects, and local high schools to reach as many "high risk" pregnant women as possible, providing free, accessible prenatal sessions, social support networks, and post-natal home visiting programs. In partnership with the education system and other social service agencies, IHSs could work to implement policies providing parenting courses at little or not cost to families. Parenting courses have been found to have a positive effect on parents' attitudes, knowledge and behaviour, as well as on child behaviour (Gottlieb et al., 1995). These positive outcomes are consistent for programs directed toward all parents interested in acquiring better parenting skills, as well as high-risk parents, suggesting that policies should include all families throughout the rostered community (Gottlieb et al., 1995). Staff could help rostered community members advocate for adequate and effective childhood programs. These could include preschool and Head Start programs, that bring groups of 3-5 year olds together in centres of school settings to offer educational programs to improve children's readiness to succeed in school, as well as health and developmental screenings, parental

involvement, and social service assistance (Gomby et al., 1995). IHSs could also be instrumental in developing policies to provide affordable, accessible child care, including prenatal support and involvement, to all families on a continuing and respite basis (Pransky, 1991; Steinhauer, 1996). Child care programs can promote child development, and free parents from their child care responsibilities so that they can work. Short- and long-term outcome studies on high-quality early childhood programs found positive effects on children's cognitive and social development, and physical health, as well as positive effects on the life outcomes of mothers (Gomby et al., 1995). It therefore makes sense for IHSs to actively support the development of such policies in an effort to promote the health of the women, children and families in their community. Local worksites might also be a potential partner for IHSs to work collaboratively with in trying to develop family-friendly work policies that provide extended parenting leave, flexible hours for pregnant women and sole support parents, and provide child-minding facilities on the premises. IHSs could also join coalitions working for changes to legislation that affect children and families - social services, health care, and financial support for lowincome families.

Given the evidence for the effectiveness of policies promoting women, children and families health, and the long-term health and social costs associated with current high risk conditions affecting children and families, it makes absolute sense for IHSs to actively work to develop and support healthy public policies (CICH, 1994; CPHA, 1997; Steinhauer, 1996). A policy strategy is a highly leveraged strategy for IHSs to focus upstream, to promote the health of its members, and to achieve greater health and social equity.

As another example, IHSs could assist in the development and implementation of policies that ensure a supply of affordable and accessible housing for its rostered members. For example, IHSs could work with municipal governments, social housing authorities, and citizens to develop innovative alternatives for public building policies. Working with financial institutions and government departments, IHSs could investigate increasing the availability of financing to make it easier for people to buy or repair houses, or set up cooperative housing structures. Community policies could be developed in collaboration with local businesses, the education system, and the stakeholders identified above to train communities in planning, building and renovating their own houses with the use of appropriate technology (Haglund et al., 1996). Reducing isolation and ensuring the inclusion of the elderly and people with physical and mental disabilities within our communities is critical to promoting the health of these groups, as well as the health of our communities (The Roeher Institute, 1994; Haldemann & Wister, 1994). In addition to strengthening community capacity to include everyone, IHSs could investigate policy options for creating local, supportive low-cost housing that permits social contact between the young, elderly, and people with disabilities. Community facilities could be built into the plans, so that the entire community benefits from, and takes part in maintaining such centres. These types of policy options have contributed to more supportive social environments, and reduced problems from lack of appropriate housing and support

services in Scandinavian and American municipalities (Haglund et al., 1996; Houben & van der Voordt, 1993; Vestbro, 1993; Silver, 1991)..

These examples are intended to demonstrate that IHSs could, and, we believe, should be concerned with public policy that addresses risk condition(s) that members of their rostered community have identified as a priority issue. It should be recognized, however, that of all risk conditions, relative income inequality has been identified as a primary determinant of health inequities (Wilkinson, 1992; Wilkinson, 1986). Relative income inequality means that after a country has achieved a certain level of Gross National Product (GNP) per capita annual income (approximately \$5,000, 1990 values), then overall health status depends more on the internal distribution of wealth than increases in income (Wilkinson, 1986). The health impacts of poverty and inequitable income distribution have been well documented (see the Canadian Public Health Association's Discussion Paper on the Health Impacts of Social and Economic Conditions, 1997 for a review of Canadian data). In Ontario, women, children and youth, seniors, people with disabilities, and aboriginal peoples are particularly vulnerable to associated health inequities (CPHA, 1997). Therefore, it is of particular importance that IHSs play a role in healthy public policy development and advocacy addressing the impact of poverty and inequitable income distribution.

Health advocacy is closely related to the strategy of healthy public policy. It is a strategy or process by which participants attempt to influence the policy decisions of governments or other authorities that exert control over the factors influencing health (Hyndman, 1997). For health promoters and health care professionals working within an IHS, particularly in the area of primary health and health-related social services, advocating for the health of one's clients and rostered community is a legitimate and valuable role. Advocacy efforts have been a critical factor in bringing about the adoption of various health promoting policies including the implementation of our National AIDS strategy, and the prior banning of print advertising of tobacco products (Kinsman, 1992; Ferrence & D'Souza, 1993).

HEALTH ADVOCACY: A VISION FOR IHS IMPLEMENTATION

IHSs could become advocates for the health of their community and for the people of Ontario in a number of ways. They could support efforts of members of their rostered community, who want to participate in the development of healthy public policy around issues that most concern them. Support could include the provision of meeting rooms, research support, administrative support in drafting reports or position papers, and staff support in mobilization, organization and outreach efforts. Furthermore, IHSs could develop stronger, united advocacy efforts through intersectoral collaboration towards a common cause. By placing health outcomes on the agenda of multiple levels of government and multiple organizations, health will no longer be thought of as the sole responsibility of the health system, but as a collective and social responsibility.

It is our position that a core IHS service should involve development, research and advocacy efforts to promote the implementation of healthy public policies. As part of this core strategy, IHSs should be required to demonstrate their involvement in a certain degree of intersectoral collaborative efforts, and their support for community mobilization and advocacy around healthy public policy issues.

CREATE SUPPORTIVE ENVIRONMENTS

Of the five health promotion action areas identified in the Ottawa Charter, the creation of supportive environments is the broadest. While the creation of supportive environments can be considered a goal or outcome of the other health promotion strategies discussed in this section, it is also a strategy that encompasses a distinct range of approaches to the promotion of individual and community health (Hyndman, 1997). The foundation of this action area is the knowledge that people's health and their relationships with their environments are inextricably linked. As a result, health cannot be separated from other goals in society (WHO, 1986). To be healthy, individuals need to live in social environments (families, communities) that are "healthy," and communities need to exist in physical environments that are "healthy." The overall guiding principle for all levels of human society (national, regional, local) should therefore be that of reciprocal maintenance -- taking care of each other, our communities and our natural environment (WHO, 1986). This principle of reciprocal maintenance extends into all human activity and organization - our work, schools, families, and communities. The strategy of creating supportive environments focuses on the settings where relationships between people and their environments occur. By seeking new, holistic ways to organize or balance these relationships, this strategy guides us towards creating social and physical environments (both natural and human-built), that are supportive of health. More specifically, it guides efforts to create environments that are safe, stimulating, satisfying, enjoyable, and sustainable, the most basic conditions required for optimal levels of health (Ottawa Charter; 1986).

The Healthy Communities movement provides an excellent example of how this strategy has been used around the world to create urban environments or communities that are supportive of health. In Ontario, the Healthy Communities Coalition is a grass roots community movement that links and facilitates partnerships among non-profit groups, neighbourhood and provincial associations to develop a broad, multi-sectoral approach to community planning (City of Toronto Healthy City Office, 1995). The 300 involved communities across the province share a strategy of creating supportive environments in order to address a diverse range of local issues. The strategy usually begins with the establishment of an intersectoral committee within the community, which includes strong representation by community residents. The committee identifies local conditions affecting the health of neighbourhoods, and "realistic" or feasible projects that will address those priority issues (Wharf-Higgins, 1992; Poland, 1996). Some of the many projects supported by communities across Ontario include self-help for homeless people, plans for sustainable development and recycling, changes to official municipal plans, improving air,

soil and water quality, and creating a community safety strategy (City of Toronto Healthy City Office, 1995). Often these projects are delegated to subcommittees, who assume responsibility for actively involving community residents and multiple sectors, and for generating funds (Hyndman, 1997). Most healthy community initiatives involve some degree of representation and support from municipal bodies, encouraging the development of innovative responses to health-related issues by local governments (Hyndman, 1997). To date, local level evaluations suggest that the strategy to create community environments that are supportive to health have achieved a number of positive environmental changes (Hyndman, 1997).

Other examples of Ontario communities that have rallied together to create environments that are supportive of health include many of the finalists for the Trillium Foundation's Caring Communities Award (Trillium Foundation, 1997). For example, in the region of Waterloo, there is a strong tradition of community building. Local businesses, police, social agencies, municipal government, and many residents from the communities of Kitchener, Cambridge and Waterloo have pulled together and found local solutions to overcome major challenges such as natural disasters, pollution crises which threatened their water supply, and economic recessions. Community resources, energy and action have resulted in among the lowest employment (7.5%) and poverty (12%) rates in the country, a 27% decrease in the rate of violent crime since 1990, and one of the lowest school drop-out rates in Ontario.

The Coalition of Community Health and Support Services in Hamilton-Wentworth provides another example of a successful strategy to create a community environment that supports health (Trillium Foundation, 1997). Today the Coalition brings together 35 non-profit or public service agencies, and a range of ethnic communities, all working toward the primary health and well-being of the region's communities and their residents. A flexible inter-service model of Team Practice, and effective use of volunteers, including volunteer training in their own neighbourhoods optimizes people's use of energy and skills. Understanding the importance of supporting family caregivers, the Coalition has consistently worked to enhance the quality of life for seniors and individuals with disabilities. This has included coordination of transportation providers and specialty services, and the establishment of an Elder Abuse Task Force. All efforts have emphasized community involvement and interservice delivery.

CREATE SUPPORTIVE ENVIRONMENTS: A VISION FOR IHS IMPLEMENTATION

IHSs could become an active participant in their area's Healthy Community initiative, or efforts similar to those in the Caring Communities described. In communities without an organized strategy in place, IHSs could initiate the establishment of an intersectoral committee, and recruit interested rostered members to get involved. IHSs could help facilitate a broad, participatory process for the community to develop a collective vision about what kind of community they want in order to optimize health and social justice.

This vision could then lead to the development of a strategic plan for the creation of those environmental conditions. IHSs could assist this process by providing statistics and user-friendly information on the health of the community, and areas of health inequity. Through sharing knowledge about the health status and needs of their entire rostered community, IHSs can help keep initiatives focused on the community as a whole. These efforts will contribute to community empowerment, enabling communities to shape public agendas and the distribution of resources, thereby achieving greater health equity and social justice (Flynn et al., 1994). IHSs could also promote their members health and well-being by dedicating staff time and resources to specific projects that address the broader determinants of health at a local level.

Another example of possible IHS involvement could be to create supportive environments that protect women from violence. Again, an intersectoral approach would be most effective in dealing with this issue, so IHSs could initiate or facilitate the development of a multi-sectoral committee with broad based community representation. IHSs could work collaboratively with women's groups, religious organizations, schools, media, and social services agencies to develop awareness and education campaigns aimed at shifting social constructions of gender and raising the issue as a community priority. IHS' staff could work in partnership to provide school programs that teach youth healthy ways to think about their sexuality and their bodies, that increase self-esteem and life skills, and that analyze issues of power imbalances in society, and in personal and work relationships. Staff could develop partnerships with local businesses and organizations to address the issue of sexual harassment. Recommended actions include the creation of empowering organizational climates based on caring and responsibility, that support diversity, and that reduce divisions based on gender (Bond, 1995). Developing crisis shelters and support networks for women within the community could be another project for IHSs to support through staff time, money, and or resources such as building space. More upstream efforts could include working to enhance women's self-sufficiency and financial security through advocating for improved child-support and employment equity policies. These outcomes could also be achieved through skill development and employment efforts.

Whatever the issues identified, a multi-dimensional, participatory strategy to create health-supporting environments is an effective way for IHSs to promote the health of their clients and rostered community. Therefore, it is our position that involvement in intersectoral, participatory, collaborative efforts to create supportive environments within their communities should be considered a core IHS area of activity, requiring dedicated financial and other resources.

SECTION IV

CORE VALUES AND THEIR IMPLICATION FOR IHSs:

As identified in Section One, the core values that characterize the field of health promotion are empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration. These same values can and should be applied to the planning, development and operation of IHSs. They will not only enrich the ethical base and understanding of health practices, they will increase the effectiveness of those who are actively working to contribute to the health and well-being of individuals and our society.

Experts in systems integration agree that the foundation for any system is not a question of structure or the function of governance, but the mission and culture of that system (Orlikoff, 1995; Ball, 1995). Therefore, the core values of empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration should be legislated as part of the IHS mandate. Each IHS should incorporate these core values into their mission and/or vision statements, which should be visibly displayed in all IHS' delivery sites. These values should guide all IHS' operations, including the design of appropriate management, funding, and accountability structures. The Canadian Hospital Association, the Ontario Nurses' Association, and experts in health system integration recognize that this position requires a paradigm shift for organizations currently operating in a medical model (Fyke, 1989; ONA, 1996; Shortell et al., 1993; Orlikoff, 1995). We join these groups in their support for a shared visioning and strategy process that would realign the culture, structure and skills of IHSs around a health promotion approach.

Many of the recommendations below support multiple values. For example, recommendations for empowering IHS' communities and clients often involve increasing public participation, while reducing social inequities often requires facilitating an intersectoral approach to address the impact of the broader determinants of health. For the sake of clarity, the following recommendation have been grouped according to the value that they primarily support. Each grouping is not intended to stand independent of the other recommendations, however. It is as an integrated framework that these recommendations will most effectively strengthen the health promoting mandate of IHSs.

EMPOWERMENT

Empowerment refers to the capacity of individuals and communities to exercise the power they require to improve their health through increasing their control over the determinants that are important to their health. IHSs should make every effort to increase the degree of control that the rostered community and the individual clients have over the determinants that effect their health.

Ensuring that rostered communities are involved in a meaningful way in the design and operations of IHSs is fundamental in this regard. Flexibility in the Ministry's legislative framework is also necessary to allow for local-level decision-making around the design and operation of each IHS. By devolving certain responsibilities for resource allocation, strategic planning, and structure to IHSs while establishing and enforcing critical provincial wide standards, the Ministry of Health can create the opportunity for community empowerment, and free IHSs to best meet their rostered community's needs.

To ensure an empowering approach to IHS design and operations, we recommend the following.

I. Two-way communication opportunities such as community forums, Town Halls, and information resources should be sponsored by the Ministry of Health, giving citizens throughout Ontario the opportunity to learn about IHSs, and enabling grassroots involvement in process of health reform.

II. IHSs should be required to develop accessible two-way information loops with their rostered members.

Information related to opportunities for involvement on IHS' decision-making bodies, committees, or projects should be provided to rostered members on a regular basis, as should information regarding IHS' priorities, strategies, structure, resource allocation, and services. Modes of communication may included Town Halls, 1-800-information phone lines, mail-outs, posters and pamphlets, and community newspapers.

Formal, accessible processes should be put in place to ensure community involvement in determining how the IHSs will address the above areas. Suggested channels of communication include sample surveys, focus groups, community consultations, and robust committee structures with high membership involvement. Whatever communication system adopted, IHSs should be accountable for ensuring their accessibility to rostered members by actively addressing systemic barriers to information such as language, literacy, physical disabilities, transportation to public events, alternative times for public events, etc..

III. The planning, development and operations process of IHSs must be open to rostered members.

This includes open Board and Annual meetings, and publicly accessible records of these meetings' minutes and resolutions. It also demands that the process and criteria used to determine membership on Governance Board and various IHS' Committees, and decisions regarding positions filled on IHS' decision-making bodies be open to public scrutiny.

- IV. We recommend the development of an IHS Report Card that will include both process indicators (e.g. IHS' operations and strategic decisions; public participation and representation; distribution and types of services and service providers) and outcome indicators (e.g. medical, lifestyle/behavioural, and socio-environmental indicators of community health status; efforts to address inequities and injustice within the rostered community).
- Consistent with the principle of empowering communities through access to information, the IHS Report Card must be made available to all rostered members as well as the general public, thereby enabling the public to make more informed choices about the IHS they wish to belong to. The IHS Report Card is one means of holding IHSs accountable to their members for its operations, costs and outcomes. It is also a tool to assist IHSs in identifying areas of strength and areas for improvement, in setting priorities and planning, and in supporting the identification of best practices and benchmarking throughout the province. To this end, the data collected and measured must be reliable, relevant, predictive, clearly operationally defined, and useful in helping IHSs continuously improve (Nelson et al., 1995).

In an effort to ensure that individual clients rostered to IHSs have as much control as possible over the determinants that effect their health, we recommend the following:

- V. The core values of empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration should be incorporated into all contracts between IHSs and their rostered clients.
- VI. A province-wide Comprehensive Client Bill of Rights for all rostered IHS' members should be developed.
- Similar in principle to Netherlands' Act on the Medical Contract, which clarifies the principal rights of patients (WHO, 1996), the Bill would enshrine clients' right to roster with an IHS that meets provincial standards of operations, core services, accessibility and quality, that upholds the core values identified above, and that protects the rights guaranteed in the Canada Health Act and the Charter of Human Rights and Freedoms.
- VII. IHSs should be held accountable for ensuring that all practitioner/client interactions are based on an autonomy-supporting partnership, consistent with the principle of empowerment.

- This requires that all service providers, enrollment processes, and services offered are sensitive to and respectful of different cultures, religions, language groups, and persons with disabilities. In support of this position, the onus should be on IHSs to provide, where appropriate, interpreters, and information materials in multiple languages (e.g. health education information, enrollment contracts).
- VIII. To increase clients' control over the factors and conditions that affect their health, the health promotion strategies outlined in Sections II and III of this paper should be incorporated within the Ministry of Health's legislative framework as core IHS services or areas of activity. All of this strategies must be developed based on the principle of empowerment, in partnership with the clients served.
- IX. Consistent with the principles of client empowerment, integration and collaboration, we support the development of an Integrated Health File for each rostered IHS' client.
- Client accessibility to their personal file must be ensured (which includes addressing potential barriers related to language, literacy, and/or physical disability), while clients' rights to privacy and confidentiality must be protected.
- X. The principle of client empowerment should be supported by ensuring that citizens are free to contract or roster with the IHS of their choice, and that the allocation of capitated funding follows the client to whichever system they choose to roster with.
- However, we recognize the limited effectiveness of this strategy for individuals, who, due to disabilities, poor health status, age, geographic location, or other disadvantages, may be prevented from choosing to roster with an alternate IHS. Therefore,
- IHSs should be held accountable for ensuring that formal structures and processes are in place to ensure client choice in service and care provider, and a "client-focus" for all services and operations.
- This latter goal could be achieved through: adopting quality enhancement programs within each IHS; establishing a client advocate body within each IHS that deals with client satisfaction and client complaints; establishing an independent, ombudsmanlike office for the province where all citizens can feel completely safe to complain about violations of their rights, neglect, abuse and inadequate services, and can bring unresolved complaints that have been brought to their IHS' client advocate body. Similar suggestions are supported by consumers groups and professional health care associations (ONA, 1996; MCCLTC, 1997).

- XI. IHSs must ensure clients have equitable access to service and to involvement on decision-making bodies and committees.
- IHSs should be held accountable for demonstrating how they are actively addressing systemic barriers to accessibility related to: race, sex, age, culture, religion, language, physical or mental disability, socio-economic status, transportation, hours of operation etc..
- While IHSs must first and foremost be concerned with meeting the health needs of their clients and community, effort should also be directed towards empowering the agencies, organizations and employees that work within the IHS. In this regard, we have the following recommendations.
 - XII. IHSs should be committed to creating work environments that are supportive of employees' physical, mental, and social health, as laid out in Health Canada's Corporate Health Model (Health Canada, 1991).
 - XIII. IHS' structures should reduce power and income hierarchies, and health providers' dependence on physicians by moving to collaborative, multidisciplinary work teams and empowering accountability structures.
 - This position is consistent with reform recommendations made by health promoters (Labonte, 1997), health systems integration experts (Gauthier, 1995; Sigmund, 1995), the Canadian Hospital Association (Fyke, 1989), the Ontario Nurses' Association (ONA, 1996), other health reform advisory bodies (WHO, 1996; Roeher Institute, 1994; Subcommittee on Primary Health Care, 1996).
 - XIV. IHSs should be held accountable for actively working to achieve parity between the salaries or financial compensation of IHS' employees working in institutional and community settings, and within health or health-related social services (Labonte, 1997).
 - XV. IHSs should be held accountable for demonstrating, to the greatest extent possible, the sustainable use of sustainable environmental resources (Labonte, 1997).

PUBLIC PARTICIPATION:

All health promotion efforts emphasize active *public participation*. The field is committed to a process that encourages and enables individuals and communities to define, analyze and act upon problems in their lives and living conditions (Labonte, 1993). It is our position that the Ministry of Health require that significant levels of public participation be incorporated into the design and ongoing operations of IHSs. This position demands that citizens have equal access to opportunities for active involvement, and that their efforts are supported with the tools, skills and resources required for meaningful participation. To support our position, we make the following recommendations.

I. IHSs should be held accountable for ensuring the meaningful participation of their rostered members in the design and planning of IHSs.

Based on Arnstein's Ladder of Citizen Participation (Arnstein, 1969), meaningful participation must go beyond consultation, where citizens have advisory or informative roles. At a minimum, we support a partnership model of participation, similar to the Finish health system in which citizens have formal influence on the health care authority through seats on a municipal health board (WHO, 1996).

This requires rostered members to hold 40% of all seats on IHS' governance boards.

Processes to protect real negotiation between citizens and other committee members must be established, and once the committee ground rules are set they should be protected from unilateral change. Making citizen representatives answerable to a clear, external constituency of the rostered population would also help to strengthen their role in decision-making (Arnstein, 1969).

As IHSs evolve, we support the move to a model of delegated power where rostered members occupy the majority of seats on the governance body.

Our vision for true community empowerment is that IHSs become not only community-based, but community owned, a position supported by experts in health systems integration and by professional health provider associations (Griffith, 1996; Warden, 1996; ONA, 1996).

II. In addition to public participation on governance structure, it is our position that IHSs should immediately begin delegating responsibility for certain projects or areas of activity to a robust committee structure, in which a majority of seats are held by rostered members.

Such committees should be contracted to undertake particular parts of IHS' projects, which might include community health assessment processes, community health advocacy functions, evaluation processes, and the selection of evaluation indicators to assess the enhancement of social health or well-being. Alongside two-way

communication systems formal community consultation processes and advisory boards, a robust committee structure will ensure public participation in multiple key areas of IHS' operations.

III. IHSs should provide opportunities for the meaningful participation of clients in development, management and day-to-day operations of services directed towards their specific needs.

For example, senior citizens should have the opportunity to participate in management and program planning activities of long-term care facilities and home care services, and multicultural services should developed in partnership with representatives from those communities.

IV. IHSs should be held accountable for establishing a process which ensures that their governance board and other key decision-making bodies are representative of the characteristics of the rostered community served.

The onus should be on IHSs to demonstrate efforts taken to remove systemic barriers to participation of disadvantaged groups, thereby facilitating the involvement of those groups not generally represented on decision-making bodies (i.e. discrimination based on ethno-cultural background, language, socio-economic status, age, sex, physical or mental disability etc.).

V. IHSs should be accountable to develop a formal process which would provide members elected or selected to sit on governance board with the tools, skills and resources necessary to for meaningful participation.

Prior skills or experience on governance bodies should not be a barrier to public participation in decision-making. Orientation training, capacity-building, and upgrading processes for Board members should address generic governance skills (e.g. resource allocation and budgeting, decision-making, group process), a broad understanding of the health system, an orientation to health promotion (e.g. the determinants of health, health promotion strategies), and an understanding of the broad-based and specific health needs of the rostered community. The United Way and the Social Planning Council provide two models for this type of comprehensive governance training process.

ADDRESSING THE IMPACT OF BROADER DETERMINANTS OF HEALTH:

Recognizing that the factors that most effect health extend beyond genetics, individual lifestyles, and the provision of a health care, part of the mandate of IHSs must be to *address the impact of broader determinants of health*. This requires IHSs to allocate money, staff and other resources to support interventions and collaborative partnerships which recognize and act upon the socio-environmental conditions that shape the world in which we live. To support this position, we recommend the following.

I. To ensure that resources are reallocated upstream and include health promoting activities that address the impact of the broader determinants of health, we support a provincial standard on the minimum budgetary proportion that IHSs must allocate to health promotion.

While further investigation is required to determine the optimal proportion of health dollars to be dedicated to health promotion, we recommend an initial minimum standard of 2-3% of the total IHS' budget be allocated specifically for Lifestyle or Behavioural approaches to health promotion (see Section II), and an additional 2-3% of the total IHS' budget dedicated to socio-environmental approaches to health promotion (see Section III).

The minimal proportion of each IHS' budget allocated towards health promotion would therefore be between 4-6%. As IHSs evolve, we recommend increasing this amount to 10% of the total budget.

These figures are distinct from, and in addition to, the money that IHSs allocate for preventive medicine, which should also be increased from current levels. These initial base amounts are consistent with the range supported by public health and health promotion research (Labonte, 1997).

II. The Ministry of Health and IHSs should work with communities in a broadbased, participatory process to identify and develop assessment indicators which reflect a broad definition of health.

Indicators should be qualitative and quantitative, and include assessments of Quality of Life and social well-being. The Ministry of Health should incorporate the responsibility of actively working to enhance these measures of health into the mandate of IHSs, and hold them accountable for the effectiveness and extent of their efforts.

III. The Ministry of Health should set aside a pocket of special funding for efforts directed towards addressing geographic issues related to the impacts of the broader determinants of health (e.g. environmental or food production issues that extend beyond the rostered community of any particular IHS).

Funding applications should be granted on a competitive basis, with applicants including, but not limited to, IHSs. Successful applicants should be those organizations or coalitions which are best able to demonstrate effectiveness, efficiency, and alignment with the core values of health promotion (empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration). Special funding is not intended to be continuous source of funding to IHSs. Pilot projects that demonstrate successful outcomes would become part of the best practices used within IHSs across the province.

REDUCING SOCIAL INEQUITIES AND INJUSTICE:

It is our firm position that IHSs be committed and accountable for working to *reduce social inequities and injustice*, such that every individual, family and community may benefit from living, learning and working in a health-supporting environment. This requires IHSs to ensure that the needs of all members of their rostered community are addressed fairly. It also demands IHSs focus special attention to the needs of those individuals who are most at risk for poor health as a result of systemic discrimination and inequity. IHSs must formally assess the impact that they are having on the health of the most marginalized members of their community. In support of this position, we make the following recommendations.

I. We support further research into the development of a risk-adjusted capitated funding formula which would serve as an effective, equitable and responsive funding mechanism for IHSs.

We believe that a risk-adjusted capitated funding system has the potential to redress current funding inequities in our system that result from historic utilization rates, and volume-driven mechanisms like fee for service.

In order to ensure that risk-adjusted capitated funding does reduce, rather than contribute to, inequities or injustices in our health system, adjustment factors must include age, sex, health status and socio-economic status.

Evidence clearly shows that an individual's socio-economic status is one of, if not the, greatest determinant of health. Without an adjustment for SES, we would create an incentive for IHSs to cream skim and/or under serve those most in need. Legislation preventing IHSs from "dumping" clients from their rosters would be inadequate protection, as IHSs could still fail to address the needs of low-income members through discriminatory practices that would be difficult for the Ministry of Health to detect or prevent.

II. We oppose the use of personal financial incentives or bonuses to IHS' providers based on savings achieved through increased efficiencies or effectiveness.

Our concern is that such a method would open the door to a profit-motive or personal gain mentality, which is contrary to the value of reducing social inequities and injustice, and of serving the clients' best-interests. We are satisfied that needs-based capitated funding, and making IHSs accountable for resource allocation decisions (within broad provincial standards for health promotion funding) provides adequate financial incentives for IHSs to adopt progressive strategies which will promote the health of their rostered communities.

We recommend that savings achieved by IHS reform be reinvested into the individual systems, so as to provide new or improved services to their rostered community.

This model of system versus personal financial incentives is similar to that used by New Zealand's independent practice associations, and has been found to be effective and well received by both clients and health providers (Malcolm, 1997).

III. To ensure that the health needs of all Ontario residents are addressed, we recommend that the Ministry of Health set clear standards against which they will hold IHSs accountable, and that the Ministry of Health become responsible for tightly monitoring and evaluating the operational and health outcomes of IHSs.

The monitoring and evaluation role of the Ministry of Health will be particularly crucial if a strategy of limited competition amongst IHSs is pursued. While limited competition may be an effective strategy to improve responsiveness to the needs of the general population served, market share provides a disincentive for IHSs to meet needs of small, high risk groups.

IV. To support the principles of reducing social inequities and injustice, we recommend that the Ministry of Health reserve special funding for programs that meet the needs of marginalized or high risk populations.

Funding applications should be granted on a competitive basis, with competition including, but not limited to, IHSs. Successful applicants should be those organizations which are best able to demonstrate effectiveness, efficiency, and alignment with the core values of health promotion (empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration). Funding should be directed towards programs that address the needs of populations who may not roster with IHSs (e.g. homeless, refugees), as well as to support the development if innovative new approaches to meeting the needs of marginalized or high risk populations within IHSs (e.g. those with mental illness, people living with AIDS, low-income single parent families).

Special funding is not intended to be continuous source of funding to IHSs. Pilot projects that demonstrate successful outcomes would become part of the best practices used within IHSs across the province.

- V. IHSs should be required to develop a formal body whose purpose would be to monitor and protect the needs of marginalized or at-risk groups within the rostered community (e.g. the Community Health Advocacy Committees used by British Columbia's regional health authorities). A majority of seats on these bodies should be held by rostered members representing the population groups in question.
- VI. The Ministry of Health should hold IHSs accountable for improvements in health status of the rostered population as a whole, as well as for actively working to narrow the gap between the health status of the general rostered community served and the health status of the most marginalized groups within that rostered community.

This requires the use of population-based health indicators, community-based health indicators, and social indicators which reflect the extent to which IHSs have reduced social inequities and injustice by addressing the special needs of high-risk populations within their rostered community.

VII.The Ministry of Health should set provincial standards for IHS' outreach efforts to encourage marginalized and at-risk populations to roster and utilize IHS' services.

The onus should be on IHSs to demonstrate the effectiveness of their efforts to overcome systemic barriers which often prevent those who are most in need from accessing health and health-related social services, and expressing their needs.

- VIII.IHSs should be required to provide care for individuals requiring immediate assistance regardless of whether or not they have an Ontario Health Card or are a rostered member of that IHS.
- IX. IHS reform should not be evaluated solely on its success in containing costs, but on wider range of societal objectives, such as enhancing the health of population (with health broadly defined), reducing health inequities and injustice, increasing efficiency, etc..

Similar objectives have been used by the World Health Organization in their evaluation of European health reform (WHO, 1996). This sort of evaluation requires that the

Ministry of Health use different accountability systems and indicators for the different approaches to promoting health outlined in section 1. Furthermore, as the different approaches to health promotion outlined in Section One are intended to achieve different outcomes, economic comparisons across health promotion approaches would be inappropriate (i.e. socioenvironmental approaches to health promotion should not be evaluated against preventive medicine or lifestyle approach to health promotion, but should be assessed according to their effectiveness at enhancing social health and wellbeing). Therefore, the Ministry of Health should develop and adopt distinct accountability systems and indicators for the different approaches to promoting health (see Section One and Appendix A for a more detailed discussion).

FACILITATE INTERSECTORAL COLLABORATION

Recognizing that the promotion of individual and community health requires an integrated effort from all sectors of our society, IHSs should be committed to *facilitating intersectoral partnerships and collaborative efforts*. Collaboration must be supported amongst the component parts of the IHS (e.g. acute care, long-term care, primary care, health promotion) in order for the IHS to effectively function as a single system. Collaboration must also be supported amongst IHSs and external organizations that have an impact on health and its determinants.

In accordance with the value of facilitating collaboration within IHSs, we have the following recommendations.

I. While the structural design of IHSs should be determined according to the unique characteristics and needs of each community, the Ministry of Health should require IHSs to adopt what has been described as a cooperative or corporate model of governance that does not represent the component (provider) parts of the system.

This model would recognize the IHS as a single not-for-profit corporation that provides and/or purchases services to a rostered community. The IHS governance board would be accountable for making decisions that are in the best interest of the population served. At the most fundamental level, "the role of governance in [IHSs] is to change organizational focus - from the parochial control of institutional assets to the stewardship of community resources coordinated and brought to bear on both the underlying and obvious community health problems" (Lerner et al., 1995). Therefore, board composition should be representative of the rostered community served. It should not, however, be representative of the components (or different provider groups) within the system. In other words, Board members would be elected or selected to solely represent the system as a whole, working in service of the rostered community. It is our position that this type of corporate or cooperative governance structure is more consistent with the goals of system integration and collaboration than

what has been described as a federated, representative, or constituency assembly model of governance.

Federated governance models, based on strategic alliances amongst several independent partners, builds fragmentation and competition into IHSs. Federated models like representative corporate models of governance, divide the loyalty and accountability of board members between the IHS' rostered community and the health organization or system component which they represent (Pointer et al, 1995). Such a structure makes full collaboration within IHSs difficult, if not impossible, as each IHS "partner" works to protect and/or advance the interests of their component part (Pointer et al, 1995). Not only would a federated model work against integration, it also presents the risk of maintaining our current health system's power imbalances, thereby preventing the reorientation of IHSs from focusing downstream on cure and treatment to more upstream health promoting strategies.

II. To ensure a fully integrated continuum of health services, IHSs should be held accountable for providing their rostered members with the following core services: primary care, secondary care, long-term care, mental health services, and health-related social services. Preventive medicine, lifestyle/behavioural-, and socio-environmental health promotion strategies should also be considered core IHS services that are provided at all levels of care.

These services should be provided to members in their homes, in the community, and in institutional settings as required, with an emphasis on providing home and community care wherever appropriate. Core services should be funded entirely through the risk-adjusted capitated funding envelope provided to each IHS based on their rostered membership (Shortell et al., 1993; Pointer et al., 1995).

III. IHSs and the Ministry of Health should be held jointly accountable for providing the citizens of Ontario with tertiary/quaternary health care.

We oppose the full integration of these services within IHSs due to their high associated costs. Instead, we support the Ontario Nurses' Association's recommendation that tertiary/quaternary care services be funded in part by IHSs which require services for their rostered members, and in part directly by the Ministry of Health (ONA, 1996).

IV. Recognizing that health and social well-being are significantly determined by broader socio-environmental factors, we join other health providers and health consultants in recommend the integration of certain health-related social services within the basket of core IHS services. In particular, we consider those social services concerned with psychosocial risk factors to be a natural and essential fit

within IHSs (e.g. children's services such as children's mental health and children's aid, and family support services).

In much of the literature on health system integration, the scope of collaboration is limited to traditional health sector services (Marriott & Mable, 1996; Leatt et al, 1995) Shortell et al, 1994). Yet without this broader, intersectoral integration, we and other health providers and health consultants believe that IHSs will be limited in their ability to enhance the health status of their rostered communities (ONA, 1996; Verlaan-Cole, 1996; Ball, 1995; Conrad & Shortell, 1996; Labonte, 1997). For health and health-related social services to effectively operate within a single system, some degree of integration between the Ministries of Health and Community and Social Services must be achieved in terms of budgets (i.e. parts of the Community and Social Services budget must be incorporated into IHS' funding envelopes), policy, macro-level systems' management, standard-setting and monitoring.

V. We recommend that the mandate or mission of the provincial health system and IHSs should include the creation of deliberate and strategic partnerships with community and social services, and other public and private sectors that provide services, organize actions, and enact policies that have an impact on the broader determinants of health (e.g. municipalities, supportive housing, community and religious groups, education, environmental protection agencies, etc.).

In order for Ontario's health system to effectively address the impact of the broader determinants of health, intersectoral collaboration beyond the scope of IHSs is needed (Verlaan-Cole, 1996; Ball, 1995; Conrad & Shortell, 1996; Lerner et al, 1995; Labonte, 1997). These efforts will require the provincial health system and IHSs to dedicate a proportion of their budgets, staff, time, and other resources to the development and support of these partnerships. To ensure their effectiveness in this area, IHSs should honour the best practices and lessons learned for building strategic, collaborative partnerships. Critical success factors can be found in health promotion literature, and from demonstration projects such as Ontario's Heart Health Action Program that depended on intersectoral collaboration (Labonte, 1997; Purdon, 1997; RBJ Health Management Associates, 1995; Zuckerman & Kaluzny, 1991).

VI.We recommend the inclusion of enhanced health and social well-being outcomes on the policy agendas of all government ministries (particularly those responsible for community and social services, education, housing, the environment, and economics). To provide steering and direction for this collaborative effort, we recommend the establishment of an interministerial committee.

This would encourage Ministries to work collaboratively towards, and to some extent be held accountable for, the improvement of health and social well-being. Representation from large affiliated agencies (e.g. Worker's Compensation Board, The United Way), might also be considered as a means of developing strategic partnerships and linkages

necessary for collaborative efforts to enhance the health and well-being of the people of Ontario.

VII.We recommend that the bulk of primary health and health promotion services be provided directly by IHSs, as these services should make up the central business of the system.

Purchasing contracts arranged with other IHSs and/or external providers must reflect the core values of the IHS, must ensure the conditions of the Comprehensive Client Bill of Rights are upheld, and must incorporate core health promotion strategies into its mode of service delivery. Whether through direct provision or purchase of core services, IHSs should be held accountable for meeting the health needs of their rostered community.

VIII.If IHS' purchasing contracts are open to competition by private, for-profit providers, we recommend that the Ministry of Health and IHSs tightly monitor and hold providers accountable to provincial and IHS standards and guidelines (e.g. issues of service accessibility, quality, and availability, alignment with the core IHS values, ensuring the conditions outlined in the Comprehensive Client Bill of Rights, etc.).

A review of European and New Zealand health care reforms found that market-oriented systems which allowed private, for-profit competition required an equal or greater amount of government activity as public, not-for-profit systems (WHO, 1996; Ham, 1997; Hornblow, 1997; Malcolm, 1997). Activity simply changed, taking on the responsibility for tight monitoring and evaluation in order to ensure that market participants are meeting the government's health care standards (WHO, 1996). Findings also revealed that anticipated cost efficiencies, improved effectiveness and decreases in waiting times for prioritized services were often unrealized (WHO,1996; Ham, 1997; Hornblow, 1997; Malcolm, 1997).

IX. IHSs should adopt remuneration methods which are consistent with the Canada Health Act, which support and promote core IHS values (i.e. empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration), and which provide an incentive for practitioners to integrate health promotion into their daily work.

CONCLUSION:

This paper has presented the case for the critical role of health promotion in IHS reform. It has clarified the holistic concept of health, and the meaning of health promotion. Three complementary approaches to promoting health were summarized, along with the need for IHSs to incorporate all three into a comprehensive approach to enhancing client and community health. The issue of IHS' accountability for promoting the health of their rostered community was discussed. A list of suggested evaluation indicators has been provided for monitoring the effectiveness and of individual/behavioural, and socioenvironmental health promotion strategies, and their impact on client and community health and social well-being.

Key health promotion strategies falling within an individual/behavioural and socioenvironmental approach to health promotion were identified, and evidence concerning their effectiveness was summarized. Visions for the integration and implementation of each strategy as a core IHS service or area of activity was described, illustrating the natural fit between IHSs and health promotion.

Most importantly, perhaps, this paper has indicated how health promotion values and principles can guide IHS reform. Practical standards and guidelines have been recommended to ensure that the values of empowerment, public participation, addressing the impact of the broader determinants of health, reducing social inequities and injustice, and facilitating intersectoral collaboration are applied to the planning, development, and operation of IHSs.

It is our position that an effective policy framework is required to ensure the development of IHSs that will promote health. It is our hope that the Ministry of Health will support this position, and will work in partnership with those who are and will be involved in IHS reform, to reorient our health system to one that promotes and enhances the health of Ontarians. For our part, we are willing and eager to be involved in all IHS initiatives that are founded on the philosophy and values of health promotion, and that incorporate health promotion strategies into the core business of IHSs.

APPENDIX A:

SUGGESTED EVALUATION INDICATORS RELATED TO AN INDIVIDUAL/ LIFESTYLE APPROACH TO HEALTH PROMOTION

<u>Increase the Proportion of the Population Practicing Healthy Eating and Physical</u> Activity Habits

- Consumption of dietary fat as a percentage of total calories (fat as a percentage of energy)
- Percentage of people aged 18 and older who participate in physical activity equivalent to daily brisk walking for at least 60 minutes (Population distribution of physical activity)
- Percentage of children in Grades K to OAC who participate in daily physical activity within the school program
- Percentage of individuals who have a body weight-for-height that puts them at increased risk to health (BMI/Body Mass Index is a measure of weight-for-height and is the weight in kilograms divided by the square of height in meters)

Reductions in Hazardous Use of Alcohol

- Average annual adult consumption of alcohol per year
- Proportion of population consuming 15 or more alcoholic drinks per week
- Population distribution of binge drinking
- Percentage of alcohol-involved motor vehicle accidents as total of all motor vehicle accidents
- Use of alcohol among young people aged 12-18
- Incidence of fetal alcohol syndrome
- Incidence of long-term health problems associated with alcohol abuse
- Incidence of harmful behaviour induced by alcohol (violence and public disorder, family violence, failure to fulfill family, work and other social roles)

Reductions in Tobacco Use and Exposure to Second Hand Smoke

- Total tobacco sales
- Proportion of current cigarette smokers
- Quitting rates of smokers
- Percentage of: men; women; young people (aged 12-19) who smoke
- Percentage of schools, workplaces and public places that are smoke-free
- Improvements in air quality
- Eliminate sales of tobacco products to minors
- Eliminate the use of tobacco products by pregnant women

Reduce Injury and Death Arising From Motor Vehicle Traffic Accidents

 Number of MVTA per 100 million kilometers traveled or per 10, 000 vehicles registered

- Death rate due to MVTA in the: 0-14; 15-24; 25-34 age groups per 100, 000
- Measurements monitoring illness and disability resulting from MVTA
- Compliance rate regarding adult restraint use (proportion of population wearing seat belts)
- Compliance rate regarding infant and child carrier use in cars

Reduce Incidence of STDs:

• Use of condoms as protection of STDs

Note: The above indicators are based on work presented in Ontario Ministry of Health, Community Health Framework Project, 1995; and Premier's Council on Health Strategy, Health Goals Committee, 1991.

SUGGESTED EVALUATION INDICATORS RELATED TO A SOCIOENVIRONMENTAL APPROACH TO HEALTH PROMOTION:

Income:

- Proportion of population living below the low income cut-off point
- Proportion of social assistance recipients
- Average employment income
- Income distribution within a population

Education:

- Population aged 15 and over with less than 9 years of education
- Adult literacy rate
- Male to female ratio of population with various educational levels
- Number of people in training programs

Work:

- Unemployment rate
- Number of lay-offs
- Proportion of population with employment benefits
- Male to female ratio of population with employment benefits
- Occupational status integration index
- Proportion of population with part-time work
- Male to female ratio of population with part-time work
- Number of reported industrial accidents

Shelter:

- Number of homeless people
- Dwellings in need of major repair
- Percentage of owner occupied dwellings

- Proportion with subsidized rent
- Average number of persons per room
- Proportion spending 30%+ on housing

Food:

- Number of people receiving food through a food bank
- Cost of a nutritious food basket

Physical Environment:

- Number of hours of moderate/poor air quality
- Frequency of poor water quality
- Public green space
- Seasonal closing of beaches
- Ultra-violet (UV) index

Health-Related Social Services:

- Number of people receiving government assistance
- Community service adequacy indicators

Violence:

- Violent crime rate
- Sexual assault rate
- Incidence of spousal/partner abuse (emotional, physical, sexual)
- Incidence of child abuse (emotional, physical, sexual)
- Incidence of adolescent abuse (emotional, physical, sexual)
- Incidence of elder abuse (emotional, physical, sexual)

Social Support:

- Proportion of single parent families
- Proportion of dysfunctional families
- Proportion of population dissatisfied with their social life
- Number of social support networks and groups within a community

Assets/Positive Attributes at an Individual and Community Level:

- Extent of public participation in IHS planning, activities and evaluation initiatives
- Volunteer participation rates
- Voter participation rates
- Community Capacity Index (Rajkumar, 1997; Jackson, 1995)
- Social Well-Being/QOL Index (Eyles, 1994)
- Mental Health Index

Note: The above indicators are based on work presented in: Ontario Ministry of Health, Community Health Framework Project, 1995; and Premier's Council on Health Strategy, Health Goals Committee.

REFERENCES

Andrews, R., and G. Franke. 1991. "The Determinants of Cigarette Consumption: A Meta-Analysis." *Journal of Public Policy and Marketing* 10: 81-100.

Arnstein, SR. 1969. "A Ladder of Citizen Participation." AIP Journal July: 216-24.

Ball, T. 1995. "Politics and Governance in Integrated Health Systems - Community Values vs. Provider Power: The Diving Forces in Health Systems Restructuring." *Managing Change* Summer: 1-17, Toronto, ON: Quantum Solutions.

Bass, JL., et al. 1993. "Childhood Injury Prevention Counseling in Primary Care Settings: A Critical Review of the Literature." *Pediatrics* 92(4): 544-50.

Bennet, EM. 1992. "Community-Based Economic Development: A Strategy for Primary Prevention." *Canadian Journal of Community Mental Health* 11(2): 11-33.

Benzeval, M., K. Judge & M. Whitehead (Eds.). 1995. *Tackling Inequalities in Health: An Agenda for Action.* London, England: King's Fund Publishing.

Beresford, S., et al. 1992. "Evaluation of a Self-Help Dietary Intervention in a Primary Care Setting." *American Journal of Public Health* 82: 79-84.

Berkman, LF., and L. Syme. 1979. "Social Networks, Host Resistance and Mortality: A Nine -Year Follow-Up Study of Alameda County Residents." *American Journal of Epidemiology* 109(2): 186-204.

Bien, TH., WR. Miller, and JS. Tonigan. 1993. "Brief Interventions for Alcohol Problems: A Review." *Addiction* 88: 315-36.

Black, D., JN. Morris, C. Smith, and P. Townsend. 1982. *Inequalities in Health: The Black Report.* Middlesex: Penguin.

Black, D., JN. Morris, C. Smith, P. Townsend, and M. Whitehead. 1988. *Inequalities in Health: The Black Report. The Health Divide*. London: Penguin.

Bond, MA. 1995. "Prevention and the Ecology of Sexual Harassment: Creating Empowering Climates." *Prevention in Human Services* 12(2): 147-73.

Botelho, RJ. and H. Skinner. 1995. "Motivating Change in Health Behavior." *Primary Care* 22(4): 565-89.

Botvin, GJ., and S. Tortu. 1988. "Preventing Adolescent Substance Abuse Through Life Skills Training." in *14 Ounces of Prevention: A Casebook for Practitioners* Eds.

Price, R., E. Cowen, R. Lorion, and J. Ramos-McKay. Washington, DC: American Psychological Association.

Boutilier, M. 1996. *The Effectiveness of Community Action in Health Promotion: A Research Perspective*. Toronto, ON: Centre for Health Promotion, University of Toronto/ParticipACTION.

Brownell, K., et al. 1980. "Evaluation and Modification of Exercise Patterns in the Natural Environment." *American Journal of Psychiatry* 137: 1540-45.

Bracht, N. (Ed.) 1990. *Health Promotion at the Community Level.* Newbury Park, California: Sage Publications.

Bruvold, WH. 1993. "A Meta-Analysis of Adolescent Smoking Prevention Programs." *Am J Public Health* 83(6): 872-80.

Bryan, AD., LS. Aiken, and SG. West. 1996. "Increasing Condom Use: Evaluation of a Theory-Based Intervention to Prevent Sexually Transmitted Diseases in Young Women." *Health Psychol* 15(5): 371-82.

Calfas, KJ., et al. 1996. "A Controlled Trial of Physician Counselling to Promote the Adapting of Physical Activity." *Preventive Medicine* 25.

Campbell, M., B DeVellis, V. Strecher, A. Ammerman, R. DeVellis, and R. Sandler. 1994. "Improving Dietary Behavior: The Effectiveness of Tailored Messages in Primary Care Settings." *American Journal of Public Health* 84(5): 783-87.

Canadian Institute of Child Health. 1994. *The Health of Canada's Children: A CICH Profile*, *2nd Edition*. Ottawa, ON: Canadian Institute of Child Health.

Canadian Public Health Association. 1997. *Health Impacts of Social and Economic Conditions: Implications for Public Policy*. Discussion Paper for the Board of Directors. Ottawa, ON: Canadian Public Health Association.

Canadian Task Force on the Periodic Health Examination. 1994. *The Canadian Guide to Clinical Preventative Health Care.* Ottawa, ON: Canada Communication Group.

City of Toronto Healthy City Office. 1995. *Toronto Healthy City News.* 3 September: Toronto, ON: City Clerk's Department.

City of Toronto Healthy City Office. 1993. *The Citizens Guide: Toronto's First State of the City Report.* June: Toronto, ON: City Clerk's Department. College of Family Physicians of Canada. 1994. *Alcohol Risk Assessment and Intervention Resource Manual for Family Physicians*. Mississauga, ON: College of Family Physicians of Canada.

Conrad, DA, and SM. Shortell. 1996. "Integrated Health Systems: Promise and Performance." *Frontiers of Health Services Management* 13(1): 3-40.

Contento, I. et al. 1995. "Special Issue: The Effectiveness of Nutrition Education and Implications for Nutrition Education Policy, Programs, and Research: A Review of Research." *Journal of Nutrition Education* 27(6): 277-422.

Cunningham, R. et al. 1992. *Promoting Better Health in Canada and the USA: A Political Perspective*. Toronto,ON: Centre for Health Promotion, University of Toronto/ParticipACTION.

Dietrich, AJ., CB. Woodruff, and PA. Carney. 1994. "Changing Office Routines to Enhance Preventive Care." *Arch Fam Med* 3: 176-83.

Dishman, RK., and J. Buckworth. 1996. "Increasing Physical Activity: A Quantitative Synthesis." *Med Sci Sports Exerc* 28(6): 706-19.

Dolan-Mullen, P., G. Ramirez, and JY. Groff. 1994. "A Meta-Analysis of Randomized Trials of Prenatal Smoking Cessation Interventions." *Am J Obstet_Gynecol* 171(5): 1328-34.

Draper, R. 1988. "Making Equity Policy." *Health Promotion* 4(2): 91-95.

Edwards, R. 1996. *Draft: Building Healthy Public Policy*. Toronto, ON: The Centre for Health Promotion, University of Toronto/PariticipACTION.

Edwards, G. (Ed.). 1994. *Alcohol Policy and the Public Good*. Oxford: Oxford University Press.

Eisen, A. 1994. "Survey of Neighborhood-Based, Comprehensive Community Empowerment Initiatives." *Health Education Quarterly* 21(2): 235-52.

Elder, JP., TL. Schmid, P. Dower, and S. Hedlund. 1993. "Community Heart Health Programs: Components, Rationale, and Strategies for Effective Interventions." *J Public Health Policy* 14(4): 463-79.

Eldrige, GD., et al. 1997. "Evaluation of an HIV Risk Reduction Intervention for Women Entering Inpatient Substance Abuse Treatment." *AIDS Education_Prevention* 9(1) Suppl.: 62-76.

European Union//Health Education Authority. 1995. *Tobacco Taxes in the European Union: How to Make Them Work for Health*. London: Health Education Authority.

Evans, RG., ML. Barer, and TR. Marmor. (Eds.) 1994. Why Are Some People Healthy and Others Not? New York: Aldine de Gruyter.

Eyles, J. 1994. *Social Indicators, Social Justice and Social Well-Being*. McMaster University Centre for Health Economics and Policy Analysis Working Paper 94-1.

Farquhar, J., M. Maccoby, P. Wood, et al. 1977. "Community Education for Cardiovascular Health." *Lancet* 1: 1192-95.

Ferrence, RG., and L. D'Souza. 1993. "Tobacco Regulation in Canada: Evaluating the Roles of Advocacy and Community Involvement in Affecting Public Policy." in *Experiences With Community Action Projects: New Research in the Prevention of Alcohol and Other Drug Problems* Eds. Greenfield, T. and R. Zimmerman. Rockville, Maryland: U.S. Department of Health and Human Services.

Fiore, MC., TE. Novotny, JP. Pierce, et al. 1990. "Methods Used to Quit Smoking in the United States." *JAMA* 263: 2760-65.

Fiore, MC., SS. Smith, DE. Jorenby, and TB. Baker. 1994. "The Effectiveness of the Nicotine Patch for Smoking Cessation. A Meta-Analysis." *JAMA* 271(24): 1940-47.

Flay, BR., and D. Burton. 1990. "Effective Mass Communication Strategies for Health Campaigns." in *Mass Communication and Public Health: Complexities and Conflicts* Eds. Atkin, C. and L. Wallack. Newbury Park, California: Sage.

Flynn, BC., DW. Ray, and MS. Rider. 1994. "Empowering Communities: Action Research through Healthy Cities." *Health Education Quarterly* 21(3) Fall: 395-405.

Freimuth, VS., and M. Kraus Taylor. 1995. *Are Mass Mediated Health Campaigns Effective? A Review of the Empirical Evidence*. Unpublished Report for the Health Communication Program, University of Maryland, College Park.

Freudenberg, N. 1985. "Health Promotion and Class." *Health/PAC Bulletin* 16(1): 24-31.

Fyke, LD. 1989. *Hospital-Based Health Promotion*. Ottawa, ON: Canadian Hospital Association.

Gauthier, A. 1995. "The Challenge of Stewardship: Building Learning Organizations." in *Learning Organizations* Eds. Chawala, S. and J. Renesch. 384-401. Portland, Oregon: Productivity Press.

Glanz, K., FM. Lewis, and BK. Rimer. 1990. "The Scope of Health Education: Parameters of a Maturing Field." in *Health Behaviour and Health Education* Eds. Glanz, FM. and BK. Rimer. San Fransisco, CA: Jossey-Bass.

Godfey, C., and A. Maynard. 1988. "Economic Aspects of Tobacco Use and Taxation Policy." *British Medical Journal* 297: 339-43.

Gomby, DS., MB Larner, CS. Stevenson, EM. Lewit, and RE. Behrman. 1995. "Long-Term Outcomes of Early Childhood Programs: Analysis and Recommendations." *The Future of Children* 5(3) Winter.

Gottlieb, LN., N. Feeley, and S. Baker. 1995. *Parent Education Programs: Effectiveness in Improving Child Mental Health*. Unpublished Report Prepared for the Child Mental Health Division, Ministry of Health and Welfare Canada. Montreal, PQ: School of Nursing, McGill University.

Gottlieb, B. 1987. "Using Social Support to Protect and Promote Health." *Journal of Primary Prevention* 8 (1-2): 49-70.

Grant, KA., PL. Lane, G. Janus, and L. Okovita. 1995. "An Evaluation of a Hospital-Based Drinking and Driving Prevention Program." *Can J Public Health* 86(2) March: 91-94.

Griffith, JR. 1996. "Managing the Transition to Integrated Health Care Organizations." *Frontiers of Health Services Management* 12(4) Summer: 4-50.

Haldemann, V. and A. Wister. 1994. "Environment and Aging." in *Aging: Canadian Perspectives*. Eds. Marshall, V. and B. McPherson. Peterborough, ON: Broadview Press.

Haglund, BJA. 1996. *Creating Healthy Environments*. Toronto, ON: The Centre for Health Promotion, University of Toronto/ParticipACTION.

Haglund, BJA., B. Petterson, D. Finer, and P. Tillgren. (Eds.). 1996. *Creating Supportive Environments for Health: Stories from the Third International Conference on Health Promotion, Sundsvall, Sweden.* Geneva: WHO.

Ham, C. 1997. "Reforming the New Zealand Health Reforms." *BMJ* 314 June: 1844-45.

Hartley, BM., and ME. O'Connor. 1996. "Evaluation of the 'Best Start' Breast-Feeding Education Program." *Arch Pediatr Adolesc Med* 150(8) August: 868-71.

Hatch, S., and I. Kickbusch, I. (Eds.). 1983. *Self-Help and Health in Europe*. Copenhagen: WHO Regional Office for Europe.

Health and Welfare Canada. 1990. *Enhancing the Provision of Preventive Services by Canadian Physicians: Report of a National Workshop.* Ottawa, ON: Health and Welfare Canada.

Health and Welfare Canada. 1990a. *Prevention Through Partnership: Collaborating for Change*. Ottawa, ON: Health and Welfare Canada.

Health and Welfare Canada. 1991. Strategy for Enhancing Preventive Practice of Health Professionals: Strategies for Today and Tomorrow. Ottawa, ON: Health and Welfare Canada.

Health Canada. 1991. *Corporate Health Model*. Health Canada and the Addiction Research Foundation.

Health Strategies Group. 1997. *Draft: Integrated Health System Policy Framework*. April, Toronto, ON: The Ontario Ministry of Health, Health Strategies Group.

Hornblow, A. 1997. "New Zealand's Health Reforms: A Clash of Cultures." *BMJ* 314 June: 1892-94.

Houben, PPJ., and TJM. van der Voordt. 1993. "New Combinations of Housing and Care for the Elderly in the Netherlands." *Neth. J. of Housing and the Built Environment* 8(3): 301-25.

Hyndman, B. 1996. *Does Self Help Help: A Review of the Literature on the Effectiveness of Self-Help Programs*. Toronto, ON: Participaction.

Hyndman, B. 1997. *Health Promotion in Action: What Works? What Needs to be Changed? Draft.* Toronto, ON: Centre for Health Promotion, University of Toronto.

Israel, B. 1985. "Social Networks and Social Support: Implications for Natural Helper and Community-Level Interventions." *Health Education Quarterly* 12 (1): 65-80.

Jackson, S. 1996. *Public Health Practitioners' Perspectives on Empowerment: Definition, Strategies and Indicators.* North York, ON: North York Community Health Promotion Research Unit.

Jacobs, D., et al. 1986. "Communitywide Prevention Strategies: Evaluation Design of the Minnesota Heart Health Program." *Journal of Chronic Disease* 39(10): 775-88.

Jaen, CR., KC. Strange, and PA. Nutting. 1994. "Competing Demands on Primary Care: A Model for the Delivery of Clinical Preventive Services." *Journal of Family Practice* 38(2): 166-71.

Jeffery, RW., CW. Gray, SA. French, WL. Hellerstedt, D. Murray, RV. Luepker, and H. Blackburn. 1995. "Evaluation of Weight Reduction in Community in Body Mass Index in the Minnesota Heart Health Program." *Int J Obesity Relat Metaab Disord* 19(1): 30-39.

- Kelly, JA., and JS. St Lawrence. 1990. "The Impact of Community-Based Groups to Help Persons Reduce HIV Infection Risk Behaviours." *AIDS Care* 2(1): 25-36.
 - Kennedy, M. 1989. "Psychiatric Hospitalization of GROWERS." Paper presented at the Biannual Conference on Community Research and Action. East Lansing, Michigan.
 - Kim, N., B. Stanton, X. Li, K. Dickersin, and J. Galbraith. 1997. "Effectiveness of the 40 Adolescent AIDS-Risk Reduction Interventions: A Quantitative Review." *J Adolesc Health* 20(3) March: 204-15.
 - King, AC., et al. 1991. "Group- Vs. Home-Based Exercise Training in Healthy Older Men and Women: A Community-Based Clinical Trial." *JAMA* 266: 1535-42.
 - King, AC. 1995. "Environmental and Policy Approaches to Cardiovascular Disease Prevention Through Physical Activity: Issues and Opportunities." *Health Education Ouarterly* 22: 499-511.
 - Kinsman, G. 1992. "Managing AIDS Organizing: 'Consultation', 'Partnership', and the National AIDS Strategy." in *Organizing Dissent: Contemporary Social Movements in Theory and Practice* Ed. Carroll, WK. Victoria, B.C.: Garamond Press.
 - Kok, G., B. van den Borne, and PD. Mullen. 1997. "Effectiveness of Health Education and Health Promotion: Meta-Analysis of Effect Studies and Determinants of Effectiveness." *Patient Educ Couns* 30(1) January: 19-27.
 - Kotler, P. 1982. *Marketing for Nonprofit Organizations*. Englewood Cliffs, New Jersey: Prentice-Hall.
 - Kotler, P., and G. Zaltman. 1975. "Social Marketing: An Approach to Planned Social Change." *Journal of Marketing* 35: 3-12.
 - Kottke, T., R. Battista, G. DeFriese, and M. Brekke. 1988. "Attributes of Successful Smoking Cessation Interventions in Medical Practice: A Meta-Analysis of 39 Controlled Trials." *JAMA* 259(19): 2883-89.
 - Kurtz, LF. 1990. "The Self-Help Movement: Review of the Past Decade of Research." *Social Work With Groups* 13(3): 101-15.
 - Kurtz, LF. 1990a. "Twelve-Step Programs." in *Working With Self-Help* Ed. Powell, TJ. Silver Spring, Maryland: NASW Press.
 - Labonte, R. 1993. *Health Promotion and Empowerment: Practice Frameworks*. Issue #3 in Health Promotion Series. Toronto, ON: Centre for Health Promotion, University of Toronto/ ParticipACTION.

Labonte R. 1997. *Health Promotion in Integrated Health Systems*. Unpublished Background Paper prepared for the Northeastern Integrated Health System Task Force.

Labonte, R. and S. Penfold. 1981. *Health Promotion Philosophy: From Victim-Blaming to Social Responsibility*. Vancouver, BC: Health Promotion Directorate, Western Regional Office.

Lalonde, M. 1974. *A New Perspective on the Health of Canadians*. Ottawa, ON: Ministryof Supply and Services.

Lasco, R., R. Curry, V. Dickson, J. Powers, S. Menes, and R. Merritt. 1989. "Participation Rates, Weight Loss, and Blood Pressure Changes Among Obese Women in a Nutrition-Exercise Program." *Public Health Reports* 104: 640-46.

Leatt, P., GH. Pink, and CD. Naylor. 1995. *Integrated Delivery Systems: Has Their Time Come in Canada?* Toronto, ON: The Hospital Management Research Unit, University of Toronto.

Lerner, WM., FL. Brown, and MD. Smith. 1995. "Governance in Integrated Delivery Systems: Serving the Public's Interest." *Frontiers of Health Services Management* 11(3) Spring: 44-46.

Lefebvre, R., and J. Flora. 1988. "Social Marketing and Public Health Intervention." *Health Education Quarterly* 15: 299-315.

Lewis, CE. et al. 1991. "The Counselling Practices of Internists." *Annals of Internal Medicine* 114: 54-58.

Linenger, JM., CV. Chesson, and DS. Nice. 1991. "Physical Fitness Gains Following Simple Environmental Change." *American Journal of Preventive Medicine* 7: 298-310.

Ling, JC. et al. 1992. "Social Marketing: Its Place in Public Health." *Annual Review of Public Health* 13: 341-362.

Link, BG., and J. Phelan. 1995. "Social Conditions: A Fundamental Cause of Disease." *Journal of Health and Social Behaviour* 80-94.

Look Good Feel Better. 1997. Handout. *The Power of Support: What Impact Does it Have on Women Living With Cancer?* CCTFA.

Lytle, LA. et al. 1996. "Changes in Nutrient Intakes of Elementary School Children Following a School-Based Intervention: Results from the CATCH Study." *Preventative Medicine* 25(4): 465-77.

Maibach, E., and DR. Holtgrave. 1995. "Advances in Public Health Communication." *Annual Review of Public Health* 16: 218-38.

Main, DS. et al. 1994. "Preventing HIV Infection Among Adolescents: Evaluation of a School-Based Education Program." *Prev Med* 23(4) July: 409-17.

Malcolm, L. 1997. "GP Budget Holding in New Zealand: Lessons for Britain and Elsewhere?" *BMJ* 314 June: 1890-92.

Mann, R., R. Smart, L. Anglin, and E. Adlaf. 1991. "Reductions in Cirrhosis Deaths in the United States: Associations with Per Capita Consumption and AA Membership." *Journal of Studies on Alcohol* 52(4): 361-65.

Mann, R., and R. Smart. 1997. *Can School Alcohol Education Programs Contribute to Declines in Drinking Among Students?* Unpublished Manuscript. Toronto, ON: Addictions Research Foundation.

Marriott, J., and AL. Mable. 1994. *Comprehensive Health Organizations: A New Paradigm for Health Care.* Kingston, ON: Queen's University, School of Policy Studies.

Marsh, A., and S. McKay. 1994. Poor Smokers. London: Policy Studies Institute.

McCrady, BS., and WR. Miller. (Eds.) 1993. *Research on Alcoholics Anonymous: Opportunities and Alternatives.* New Brunswick, NJ: Rutgers Centre of Alcohol Studies.

Medvene, LJ. 1990. *Selected Highlights of Research with Self-Help Groups*. Unpublished Report.

Metro Consumers for Community-Based Long-Term Care. 1996. "Arm's-Length Complaints Procedure: The Only Sure Way to Get Quality Control." *Network News* 3 June.

Metro Consumers for Community-Based Long-Term Care. 1997. "Complaints Procedure Committee." *Network News* 6 June.

Minkler, M. 1985. "Building Supportive Ties and Sense of Community Among the Inner-City Elderly: The Tenderloin Senior Outreach Project." *Health Education Quarterly* 12(4): 303-14.

Minkler, M. 1990. "Improving Health Through Community Organization." in *Health Behavior and Education: Theory, Research and Practice*. Eds. Glanz, K., FM. Lewis, and BK. Rimer. San Fransisco, CA: Jossey-Bass.

Minker, M. 1992. "Community Organization Among the Elderly Poor in the United States: a Case Study." *International Journal of Health Services* 22(2): 303-316.

Minkler, M. and K. Cox. 1980. "Creating Critical Consciousness in Health: Applications of Freire's Philosophy and Methods to the Health Care Setting." *Int J Health Serv* 10: 31-22.

Mintz, J. 1989. "Social Marketing: New Weapon in an Old Struggle." *Health Promotion* (Health and Welfare Canada) 27(3): 6-12.

Mittlemark, MB., MK. Hunt, GW. Heath, and TL. Schmid. 1993. "Realistic Outcomes: Lessons From Community-Based Research and Demonstration Programs for the Prevention of Cardiovascular Diseases." *J Public Health Policy* 14(4): 437-62.

Moskowitz, JL. 1989. "The Primary Prevention of Alcohol Prevention: A Critical Review of the Research Literature." *Journal of Alcohol Studies* 50(1): 54-88.

Mustard, JF., and JW. Frank. 1991. *The Determinants of Health*. Toronto, ON: Canadian Institute for Advanced Research, CIAR Population Health Program Publication #5.

Nader, P. et al. 1989. "A Family Approach to Cardiovascular Risk Reduction: Results from the San Diego Family Health Project." *Health Education Quarterly* 16: 229-44.

Nelson, EC., PB. Batalden, SK. Plume, NT. Mihevc, and WG. Swartz. 1995. "Report Cards or Intrument Panels: Who Needs What?" *Journal on Quality Improvement* 21(4) April: 155-66.

Noack. H., and D. McQueen. 1988. "Health Promotion Indicators." *Health Promotion* 3(1): 1-125.

Ockene, JK. 1987. "Physician-Delivered Interventions for Smoking Cessation: Strategies for Increased Effectiveness." *Prev Med* 16(5): 723-37.

Oldenberg, B., N. Owen, M. Parle, and M. Gomel. 1995. "An Economic Evaluation of Four Work Site Based Cardiovascular Risk Factor Interventions." *H Educ Q* 22(1) February: 9-19.

Ontario Injury Prevention Resource Centre. 1996. *Best Practice Programs for Injury Prevention*. Toronto, ON: Ontario Injury Prevention Resource Centre, Ontario Public Health Association.

Ontario Ministry of Health. 1994. *Partners in Action: Ontario's Substance Abuse Strategy*. Toronto, ON: Queen's Printer for Ontario.

Ontario Nurses' Association. 1996. *Dialogue on Health Reform: A Vision for Saving Medicare*. Summer. Toronto, ON: Ontario Nurses' Association.

Ontario Public Health Association. 1993. *Shaping the Shift to Community Health: Policy Paper, November 1993.* Passed at the Annual General Meeting in Toronto, ON.

Ontario Task Force on the Primary Prevention of Cancer. 1995. *Recommendations for the Primary Prevention of Cancer*. Toronto, ON: Queen's Printer for Ontario.

Orleans, CT. et al. 1991. "Self-Help Quit Smoking Interventions: Effects of Self-Help Materials, Social Support Instructions, and Telephone Counselling." *Journal of Consulting and Clinical Psychology* 59: 439-48.

Orlikoff, JE. 1995. "Mission-Based Governance." *Frontiers of Health Services Management* 11(3) Spring: 47-50.

Osganian, SK. et al. 1996. "Changes in the Nutrient Content of School Lunches: Results from the CATCH Eat Smart Food Service Intervention." *Preventative Medicine* 25(4): 400-12.

Pederson, LL., N. Baldwin, and NM. Lefcoe. 1981. "Utility of Behavioral Self-Help Manuals in a Minimal-Contact Smoking Cessation Program." *Int J Addict* 16(7): 1233-39.

Pekurinen, M. 1991. *Economic Aspects of Smoking: Is There a Case for Government Intervention in Finland?* Unpublished. PhD Thesis, University of New York. New York, NY.

Pointer, DD., JA. Alexander, and HS. Zuckerman. 1995. "Loosening the Gordian Knot of Governance in Integrated Health Care Delivery Systems." *Frontiers of Health Services Management* 11(3) Spring: 3-37.

Poland, B. 1996. "Knowledge Development and Evaluation in, of and for Healthy Community Initiatives. Part I: Guiding Principles." *Health Promotion International* 11(3): 237-248.

Posner, BM., et al. 1994. "Nutrition Policies and Interventions for Chronic Disease Risk Reduction in International Settings." *Nutrition Reviews* 52(5): 179-187.

Powell, T.J. (Ed.). 1994. *Understanding the Self-Help Organization: Frameworks and Findings*. Thousand Oaks, California: Sage.

Premier's Council on Health Strategy. 1991. *Towards Health Outcomes: Goals 2 and 4: Objectives and Targets.* Toronto, ON: Premier's Council on Health Strategy.

Pransky, J. 1991. Prevention: The Critical Need. Springfield, MO: Burrell Foundation & Paradigm Press.

Purdon, L. 1997. "Integrated Service Delivery -- A SETO Perspective." South East Toronto Project: Insights into Community Health Initiatives, 1(1) February: 1-11.

Rachlis, M., and C. Kushner. 1989. *Second Opinion: What's Wrong With Canada's Health-Care System and How to Fix It.* Toronto, ON: Collins.

Raeburn, J. 1996. How Effective is Strengthening Community Action as a Strategy for Health Promotion? An Empowerment/Community Development Perspective.

Toronto, ON: The Centre for Health Promotion, University of Toronto/ParticipACTION.

Raeburn, J. and I. Rootman. 1989. "Towards an Expanded Health Field Concept: Conceptual and Research Issues in a New Age of Health Promotion." *Health Promotion International* 3: 383-92.

Rajkumar, E. 1997. *Outcomes of Community Development in a Community Health Centre Setting: Towards an Acceptable Evaluation Framework.* Unpublished Report for the Health Promotion Program, Faculty of Medicine at the University of Toronto. Toronto. ON.

RBJ Health Management Associates. 1995. *The Heart Health Action Program: Final Evaluation Report, December 1995*. Toronto, ON: Heart Health Resource Centre & Ontario Public Health Association.

RBJ Health Management Associates. 1995. *Heart Health Action Program Preliminary Report: Lessons Learned from the Early Development of the Five Demonstration Projects*. Toronto, ON: Heart Health Resource Centre & Ontario Public Health Association.

Redman, S., EA. Spencer, and RW. Sanson-Fisher. 1990. "The Role of Mass Media in Changing Health-Related Behaviour: A Critical Appraisal of Two Models." *Health Promotion International* 5(1): 85-101.

Riessman, F., and D. Carroll. 1995. *Redefining Self-Help: Policies and Practice*. San Francisco: Jossey-Bass.

The Roeher Institute. 1994. *Health Care Planning Innovations Project: Literature Review*. October, Toronto, ON: The Roeher Institute for Health Care Planning Innovations Project.

Rogers, J. 1989. *Mutual Aid as a Mechanism for Health Promotion and Disease Prevention*. Ottawa, ON: Health and Welfare Canada.

- Romeder, JM., et al. 1990. *The Self-Help Way Mutual Aid and Health*. Ottawa, ON: Canadian Council on Social Development.
- Rootman, I., and L. Hershfield. 1994. "Health communication research: broadening the scope." *Health Communication* 6(1): 71-72.
- Rootman, I., and M. Goodstadt. 1996. *Health Promotion and Health Reform in Canada.* A Position Paper by the Canadian Consortium for Health Promotion Research, Toronto, ON.
- Ross, DP., and PJ. Usher. 1986. *From the Roots Up: Economic Development as if Community Mattered*. Croton-on-Hudson, New York: The Bootstrap Press.
- Rundall, TG., and WH. Bruvold. 1988. "A Meta-Analysis of School-Based Smoking and Alcohol Use Prevention Programs." *Health Education Quarterly* 15(3): 317-34.
 - Sanchez-Craig, M., R. Davila, and G. Cooper. 1996. "A Self-Help Approach for High-Risk Drinking: Effect of an Initial Assessment." *J Consult Clin Psychol* 64(4): 694-700.
- Schaalma, HP. et al. 1996. "Planned Development and Evaluation of AIDS/STD Education for Secondary School Students in The Netherlands: Short-Term Effects." *Health Education Quarterly* 23(4) November: 469-87.
- Schooler, C. 1995. *Physical Activity Interventions: Evidence and Implications*. Unpublished Report for Standford Centre for Research in Disease Prevention, Standford University School of Medicine. Palo Alto, CA.
- Self-Help Resource Centre for Metro Toronto. 1996.
- Shah, CP. 1994. *Public Health and Preventive Medicine in Canada, Third Edition*. Toronto, ON: University of Toronto.
- Shea, S., and CA. Basch. 1990. "Review of Five Major Community-Based Cardiovascular Disease Prevention Programs. Part 2: Intervention Strategies, Evaluation Methods, and Results." *Am J Health Promo* 4(4): 279-87.
- Sheehan. M., C. Schonfeld, R. Ballard, F. Schofield, J Najman, and V. Siskind. 1996. "A Three Year Outcome Evaluation of a Theory Based Drink Driving Education Program." *J Drug Educ* 26(3): 295-312.
 - Shortell, SM., RR. Gillies, DA. Anderson, JB. Mitchell, and KL. Morgan. 1993. "Creating Organized Delivery Systems: The Barriers and Facilitators." *Hospital & Health Services Administration* 38(4) Winter: 447-66.

Shortell, SM., RR. Gillies, and DA. Anderson. 1994. "The New World of Managed Care: Creating Organized Delivery Systems." *Health Affairs* (Winter).

Shortell, SM, RR. Gillies, DA. Anderson, JB. Mitchell, and KL. Morgan. 1996. *Remaking Healthcare in America: Building Organized Delivery Systems.* San Fransisco, CA: Jossey-Bass.

Sibthorpe, B., D. Flemling, and J. Gould. 1994. "Self-Help Groups: A Key to HIV Risk Reduction for Injection Drug Users." *Journal of Acquired Immune Deficiency Syndrome* 7(6): 592-597.

Sigmond, RM. 1995. "Back to the Future: Partnerships and Coordination for Community Health." *Frontiers of Health Services Management* 11(4) Summer: 5-36.

Silagy, C., D. Mant, Fowler, and M. Lodge. 1994. "Meta-Analysis of Efficacy of Nicotine Replacement Therapies in Smoking Cessation." *Lancet* 343(8890): 139-42.

Silver, H. 1991. "State, Market, and Community: Housing Co-operatives in Theoretical Perspective." *Neth. J. of Housing and the Built Environment* 6(3): 185-203.

Skinner, H. 1993. "Early Identification of Addictive Behaviours Using a Computerized Lifestyle Assessment." in *Addictive Behaviours Across the Lifespan: Prevention, Treatment and Policy Issues.* Eds. Baer, JS., GA. Mariatt, and RJ. McMahon. Newbury Park, CA: Sage.

Skinner, H., and K. Bercovitz. 1996. *Person-Centred Health Promotion*. Toronto, ON: The Centre for Health Promotion, University of Toronto/ParticipACTION

Smith Ulione, M. 1997. "Health Promotion and Injury Prevention in a Child Development Centre." *J Pediatr Nurs* 12(3) June: 148-54.

Sobell, LC. et al. 1993. "What Triggers the Resolution of Alcohol Problems Without Treatment." *Alcoholism: Clinical and Experimental Research* 17(2): 217-24.

Stall, R., and P. Beirnacki. 1986. "Spontaneous Remission from the Problematic Use of Substances: An Inductive Model Derived from a Comparative Analysis of the Alcohol, Opiate, Tobacco, and Food/Obesity Literatures." *The International Journal of the Addictions* 21(1): 1-23.

Steinhauer, PD. 1996. *Why Voices for Children is Needed*. September: Toronto, ON: Voices for Children, Coalition for Children, Families and Communities.

Strecher, V., M. Kreuter, DJ. Den Boer, S. Kobrin, H. Hospers, and C. Skinner. 1994. "The Effects of Computer-Tailored Smoking Cessation Messages in Family Practice Settings." *The Journal of Family Practice* 39(3) September: 262-68.

- Sweanor, D. 1991. *Canadian Tobacco Tax Project 1985-91*. Ottawa, ON: Non-Smokers Rights Association.
- Tobler, NS. 1994. *Meta-Analysis of Adolescent Drug Prevention Programs*. A Dissertation Submitted to the State University of New York at Albany in Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy, School of Social Welfare.

Tones, K. 1986. "Health Education and the Ideology of Health Promotion: A Review of Alternative Approaches." *Health Education Research* 1(1): 3-12.

Toseland, RW., CT. Rossiter, and M. Labrecque. 1989. "The Effectiveness of Three Group Intervention Strategies to Support Family Caregivers." *American Journal of Orthopsychiatry* 59: 420-29.

Toseland, RW., CM. Rossiter, T. Peak, and P. Hill. 1990. "Therapeutic Processes in Peer Led and Professionally Led Support Groups for Caregivers." *International Journal of Group Psychotherapy* 40: 279-303.

Townsend, J. 1988. *Price, Tax and Smoking in Europe*. Copenhagen: WHO.

Townsend, J., P. Roderick, and J. Cooper. 1994. "Cigarette Smoking by Socioeconomic Group, Sex and Age: Effects of Price, Income and Health Publicity." *British Mdical Journal* 309: 92-7.

Trillium Foundation. 1997. *The Caring Communities Award: The Newsroom*. Trillium Foundation Website. October: Toronto.

Trojan, A. 1989. "Benefits of Self-Help Groups: A Survey of 232 Members from Sixty-Five Disease Related Groups." *Soc Sci Med* 29(2): 225-32.

Utz, SW., GF. Shuster, E Merwin, and B. Williams. 1994. "A Community-Based Smoking-Cessation Program: Self-Care Behaviors of Success." *Public Health Nurs* 11(5): 291-99.

Verlaan-Cole, L. 1996. "Integrated Health Systems: From Vision to Action," *Managing Change*. Summer: 1-18. Toronto, ON: Quantum Solutions.

Vestbro, DU. 1993. "The Study of Collective Housing: A Swedish Perspective." in *The Meaning and Use of Housing: International Perspectives, Approaches and their Applications*. Ed. Arias, EG. Aldershot: Avebury.

Viswesvaran, C. and FL. Schmidt. 1992. "A Meta-Analytic Comparison of the Effectiveness of Smoking Cessation Methods." *J Appl Psychol* 77(4): 554-61.

Voices for Children, Coalition for Children, Families and Communities. 1996. *Fact Sheets #1-6.* Toronto, ON: Voices for Children, Coalition for Children, Families and Communities.

Voices for Children, Coalition for Children, Families and Communities. 1996. *Community Action Guide #1.* Toronto, ON: Voices for Children, Coalition for Children, Families and Communities.

Wagman, LM. 1993. "A Health Department's Response to AIDS. Condomania: A public Education Intervention." *Canadian Journal of Public Health* 84(1) Suppl.: S62-S65.

Walker, R. 1991. *Testimony on Behalf of the Canadian Council on Children and Youth to the Standing Committee on Health and Welfare*. Ottawa, ON: Social Affairs, Seniors and the Status of Women: Sub-Committee on Poverty.

Wallace, P., and A. Haines. 1984. "General Practitioner and Health Promotion: What Patients Think." *British Medical Journal* 289: 663-668.

Wallerstein, N. 1993. "Empowerment and Health: The Theory and Practice of Community Change." *Community Development Journal* 28(3) July: 218-27.

Warden, G. 1996. "Key Factors in the Transition to IHCOs." *Frontiers of Health Services Management* 12(4) Summer: 53-56.

Wasley, MA., SE. McNagny, VL. Phillips, and JS. Ahluwalia, 1997. "The Cost-Effectiveness of the Nicotine Transdermal Patch for Smoking Cessation." *Pre Med* 26(2): 264-70.

Weitz, D. 1988. "Notes of a 'Schizophrenic Shitdisturber'." in *Shrink Resistant: The Struggle Against Psychiatry in Canada*. Eds. Burstow, B. and D. Weitz. Vancouver, BC: New Star Books.

Wharf-Higgins, J. 1992. "The Healthy Community Movement in Canada." in B. Wharf (Ed.). *Communities and Social Policy in Canada.* Toronto: McClelland and Stewart.

WHO. 1978. Alma Ata. Primary Health Care. Geneva: WHO.

WHO. 1986. *The Ottawa Charter on Health Promotion*. Copenhagen: World Health Organization Regional Office for Europe.

WHO. 1992. *Twenty Steps for Developing a Healthy Cities Project.* Geneva: WHO Europe.

WHO. 1996. *European Health Care Reforms: Analysis of Current Strategies*, *Summary.* Copenhagen, Denmark: WHO Regional Office for Europe.

Wilk, AI., NM. Jensen, and TC. Havighurst. 1997. "Meta-Analysis of Randomized Control Trials Addressing Brief Interventions in Heavy Alcohol Drinkers." *J Gen Intern Med* 12(5): 274-83.

Wilkinson, RG. 1984. "The Epidemiological Transition: From Material Scarcity to Social Disadvantage?" *Daedalus* Journal of the American Academy of Arts and Sciences, Health and Wealth, Fall: 61-78.

Williams, GC., EL. Deci, and RM. Ryan. 1996. *Building Partnerships by Supporting Autonomy: Promoting Maintained Behavior Change and Positive Health Outcomes.* Unpublished Manuscript.

Wilson, MG. 1991. "Cholesterol Reduction in the Workplace and in Community Settings." *Journal of Community Health* 16(1): 49-65.

Wilson, DMC. et al. 1988. "A Randomized Trial of Family Physician Intervention for Smoking Cessation." *Journal of the American Medical Association* 260(11): 1570-75.

Zablocki, E. 1996. "Improving Community Health Status: Strategies for Success." *The Quality Letter* February.

Zuckerman, HS., and AD. Kaluzny. 1991. "The Management of Strategic Alliances in Health Service." *Frontiers in Health Services Management* 7 Spring: 3-23.